ANALYZING THE PRECONDITIONS FOR SUCCESSFUL COMMUNITY-BASED ECOTOURISM

BY CONDUCTING A CASE STUDY META-ANALYSIS



Source: Steppes Travel 2020

MSc Thesis Sustainable Development Simone Lassauw 6629059 October 2020



"THE ROAD TO SUCCESSFUL COMMUNITY-BASED ECOTOURISM"

MSc Thesis (45 ECTS) Student: Simone Lassauw Student No.: 6629059 Contact details: s.m.t.lassauw@students.uu.nl

Supervisor: dr. Frank van Laerhoven 2nd reader: dr. Carel van Dieperink

Master: Sustainable Development Track: Earth System Governance Faculty of Geoscience Utrecht University

Picture cover page © Steppes Travel 2020. Retrieved from: <u>https://www.steppestravel.com/destinations/central-america/belize/toledo-district/</u>

PREFACE

It is my pleasure to present to you my MSc thesis about the preconditions for successful communitybased ecotourism. It has been written to fulfill my Master's program Sustainable Development – track Earth System Governance at Utrecht University. However, I am more proud of the contribution this research may provide for the 'outside world', in particular the community-based ecotourism industry. From February 2020 until October 2020, I've been working on this research to explore and analyze the preconditions for the success of community-based ecotourism enterprises and their involved communities. The reason for this subject was raised from my interests in the field of nature conservation, socio-economic development, and environmental governance.

My search for a good match between my interests and supervisor was the reason why I contacted dr. Frank van Laerhoven to assist me during my thesis research. His expertise concerning environmental governance and interests in common-pool resources and socio-ecological systems provided suitable insights to apply to the concept of community-based ecotourism. Together we discussed the opportunities to narrow down my interests which resulted in the subject of this research. To note, one of his funny sayings has stayed with me: "it is not about how many attributes a closet has, but on how well the closet is built" referring to my research. With this in mind, my thesis got automatically structured during the different research phases.

I want to express my gratitude to Frank, for his sublime guidance, comments, and enthusiasm throughout the whole process. I am very pleased with the consultation meetings we had and the support you provided. I would like to thank the people who were willing to participate in the interviews and survey and made time for me, without the insights you provided I couldn't validate my results. Hopefully, better times are coming soon to the tourism industry. Furthermore, I would like to thank dr. Carel van Dieperink, my 2nd reader, for making time in his agenda on short notice so I could graduate in October. Lastly, I want to thank my mom for her massive support and her belief in me, this has helped me a lot.

I hope you enjoy reading it!

Simone Lassauw Schin op Geul, October 2020

"I want to dedicate this thesis to my dad, who was there in spirit \star "

ABSTRACT

Community-based ecotourism (CBET) is a promising approach to find a balance between nature conservation and socio-economic development. However, CBET from the community's perspective is little understood. Besides, it is still uncertain which preconditions contribute to successful CBET enterprises. In the existing literature, the preconditions are not often explored universally to learn from success stories elsewhere. Therefore, this research used the following research question as a guideline: *What factors contribute to the success of community-based ecotourism?*

A theoretical framework was developed containing 19 critical success factors (CSFs) clustered in six categories. These factors were derived from claims in the existing literature and were applied to 50 CBET case studies to gather insights concerning the relative importance per CSF, the six categories, and additional findings of the CBET definition, missing factors, and configurations. A triangulation of methods was used to create a set of preconditions based on the relative importance of the CSFs. The meta-analysis was used to determine the relative importance by using a coding process. The survey and interviews were conducted as additional methods to validate the outcomes of the coding process.

The results of the relative importance were defined by three components: (1) the frequency of the CSF provided an estimation of the 'relative weight' of each factor; (2) the frequency of codes provided the 'relative direction' indicating if the importance is influenced positively or negatively by each factor; and (3) the frequency of the values indicated the 'relative direction' as well. Eventually, the relative importance was ranked by using different scales in a synthesis.

The main results indicated the factors *accountability, autonomy, alternative land use resources, infrastructure,* and *funding* as relatively important, followed by the other 12 CSFs, and these were validated by the survey and interview responses. The factors *project plans as working document* and *understanding of relevant state policies* were not validated as an important factor and therefore, these two factors were excluded in the set of preconditions.

These findings resulted in the set of preconditions, i.e. set of hypotheses. This set of precondition indicates which CSFs are necessary for developing and sustaining successful CBET. This provided insights to create better conditions for local people and our planet in the context of CBET. Further research is necessary to test the generated hypotheses, to calculate the correlation between the factors and the actual success. But more importantly, the road to successful CBET presented in this research is a reliable first step.

Keywords: community-based ecotourism, critical success factors, and the set of preconditions.

TABLE OF CONTENTS

| List of Figures and Tables | 7 |
|--|----|
| 1. Introduction | 8 |
| 1.1. Background and problem definition | 8 |
| 1.2. Research objective, and main question | 9 |
| 1.3. Research framework and sub-questions | |
| 1.4. Scientific and societal relevance | |
| 1.5. Scope of the research | |
| 1.6. Outline | |
| 2. Theoretical Background | |
| 2.1. Concepts and Definitions | |
| 2.1.1. Main concepts and definitions | |
| 2.1.2. Community-based ecotourism criteria | |
| 2.2. Building a theoretical framework | |
| 2.2.1. Final clustering | |
| 2.3. Theoretical framework | 20 |
| 3. Research Methods | 21 |
| 3.1. Case study meta-analysis | |
| 3.2. Case study selection | |
| 3.2.1. Case study selection criteria | |
| 3.2.2. Case study characteristics | 23 |
| 3.3. Operationalization | 24 |
| 3.3.1. The critical success factors | 24 |
| 3.3.2. Coding process | 25 |
| 3.4. Data collection | |
| 3.4.1. Desk research | |
| 3.4.2. Survey | |
| 3.4.3. Semi-structured interviews | |
| 3.5. Analysis | |
| 3.5.1. Case study meta-analysis | |
| 3.5.2. In-depth analysis | 30 |
| 3.5.3. Synthesis | |
| 3.5.4. Set of preconditions | |
| 4. Results | |
| 4.1. Case study meta-analysis | |
| 4.1.1. Distribution per component | |

| 4.1.2. Overview of the frequency |
|---|
| 4.1.3. Overview of the codes |
| 4.1.4. Overview of the values |
| 4.1.5. Additional findings |
| 4.2. In-depth analysis |
| 4.2.1. The outcomes per category |
| 4.2.2. CBET definition in practice |
| 4.3. Synthesis |
| 4.3.1. Set of preconditions for successful cbet |
| 5. Discussion |
| 5.1. Innovative aspects |
| 5.1.1. Theoretical implications |
| 5.1.2. Methodological aspects |
| 5.2. Unexpected results |
| 5.3. Limitations |
| 5.4. Future research |
| 5.5. Policy recommendations |
| 6. Conclusion61 |
| References |
| Appendix I |
| Appendix II74 |
| Appendix III |

LIST OF FIGURES AND TABLES

Figure 1. Research framework. Figure 2. Scope of literature review. Figure 3. Steps to conduct a literature review. Figure 4. Theoretical framework. Figure 5. Boxplot component 1 - Frequency distribution of the CSFs. No outliers were found. Figure 6. Boxplot component 2- Frequency distribution of the codes. The black dot indicates an outlier. Figure 7. Boxplot component 3 - Frequency distribution of the values. The black dots indicate two outliers. Figure 8. Overview of frequency - Total number of case studies wherein a CSF was found. Figure 9. Overview of frequency per code – How often the CSFs were found per code (1, 0, -1, X). Figure 10. Outcomes of case study meta-analysis – Frequency and frequency per code – Category 1. Figure 11. Outcomes of case study meta-analysis – Frequency and codes – Category 2. Figure 12. Outcomes of case study meta-analysis – Frequency and frequency per code – Category 3. Figure 13. Outcomes of case study meta-analysis – Frequency and frequency per code – Category 4. Figure 14. Outcomes of case study meta-analysis – Frequency and frequency per code – Category 5. Figure 15. Outcomes of case study meta-analysis – Frequency and frequency per code – Category 6. Figure 16. Infographic - set of preconditions for the success of CBET.

Table 1. List of general criteria for analyzing the definition of community-based ecotourism.

Table 2. Final clustering of the CSFs per category, including their description and references.

Table 3. Case Study Selection Criteria.

Table 4. Operationalization.

Table 5. Overview of the used codes and their explanation.

Table 6. Overview of representative case studies participated in an online interview (or via email).

Table 7. The importance, influence, and validity scales used for filling in the synthesis Table.

Table 8. Overview of the relative frequency (%) per code.

Table 9. Overview of the relative frequency (%) per value.

Table 10. Overview of missing factors.

Table 11. Noticed interlinkages between the CSFs useful for identifying configurations.

Table 12. Overview of the number of case studies wherein the CBET criteria were found.

Table 13. Overview of survey responses regarding the criteria for CBET.

Table 14. Synthesis.

Table 15. Set of preconditions for successful community-based ecotourism (CBET).

Table 16. Summary of explanatory factors found in literature reviews.

Table 17. The selected case studies for the case study meta-analysis, based on the Case Study Selection Criteria.

1. INTRODUCTION

1.1. BACKGROUND AND PROBLEM DEFINITION

One of the biggest concerns is that biodiversity loss leads to crossing one of the planetary boundaries and that poor countries are dealing with the biggest consequences (Lee & Neves, 2009; Steffen et al., 2015; United Nations, 2019). The linkages between poverty, inequality, and environmental degradation have been a serious problem for a long time and therefore a well-balanced solution is needed (Baloch et al., 2020; Barnes et al., 2017; Bennett et al., 2015; Keeble, 1988). Concerning this, several scholars suggest potential approaches such as governing the commons, stakeholder collaboration, sustainable livelihoods, and collective action (Baral & Stern, 2009; Barnes et al., 2017; Hasan et al., 2020; Amanda Lee Stronza, 2010).

Community-based ecotourism (CBET) is suggested as a promising approach for finding a balance between nature conservation and socio-economic development and is as a specific type of community-based enterprises (CBE) (Anup, 2017; Asadi & Kohan, 2011; Ortega-Álvarez & Calderón-Parra, 2020; A. L. Stronza et al., 2019). CBE has emerged from the field of entrepreneurship and sustainability (Berkes & Davidson-Hunt, 2010; Gurău & Dana, 2018; Ortega-Álvarez & Calderón-Parra, 2020; Peredo et al., 2006). They focus on a broader goal instead of being based on utilitarian economics. Their view includes political, social, cultural, environmental, economic, and sustainable perspectives and they focus on opportunities, networks, skills, and knowledge to eventually overcome the challenges of limited resources, institutional constraints, and biodiversity loss (Berkes & Davidson-Hunt, 2010; Gray et al., 2018; McIntosh & Renard, 2010).

CBET is a form of ecotourism linked to community-based development (Denman, 2001). The International Union for the Conservation of Nature (IUCN) defines ecotourism as "environmentally responsible travel and visitation to relatively undisturbed natural areas, to enjoy and appreciate nature that promotes conservation, has low visitor impact, and provides for beneficially active socio-economic involvement of local populations" (Haenn et al., 2016, p. 382). Therefore, ecotourism is a development strategy but is missing the capacity-building aspect and planning from the local perspective (Koens et al., 2009).

CBET is therefore seen as a promising solution striving for nature conservation and socioeconomic development. The only difference between ecotourism and CBET is that ecotourism is managed by the private sector and assisted by the public sector where needed while CBET is managed by the local communities who control the development of ecotourism through community involvement and participation (Denman, 2001; Snyman, 2017). Through CBET management, the local people get access to the ecotourism benefits rather than experiencing the costs (Denman, 2001; Smit, n.d.). Furthermore, CBET can be seen as a source of conservation funding because the revenues can be used for conservation efforts such as protecting habitats and supporting sustainable trekking. This kind of funding can stimulate local community projects (Grigg, 2016; Gumede & Nzama, 2019; Stronza et al., 2019).

However, organizing ecotourism is difficult because of its challenges in realizing ecological and socio-economic development. The focus is rarely on both aspects (Stronza et al., 2019). CBET is even more difficult to achieve because it is easy to start but hard to sustain which results in unsuccessful CBET (Hamzah & Mohamad, 2012). According to Farrelly (2011), small ecotourism businesses have a high failure rate of 80% even under good market conditions, accessibility, and support. And not always are economic benefits generated for the local people or strong conservation actions implemented (Ohl-Schacherer et al., 2008). Furthermore, ecotourism from the community's perspective is little understood,

and more focus is required on community participation, ownership, management, and empowerment to achieve successful sustainable tourism (Sakata & Prideaux, 2013; Stone, 2015).

By understanding the concept of CBET from the community's perspective, the goal of CBET can be achieved which is to protect and conserve biodiversity and produce socio-economic benefits for local communities (Kiss, 2004). This contribution is linked to the term 'collective action' and is defined as several people working together in a group project to achieve a common objective and share the resulted benefits (Lettinga et al., 2020). Getting communities to engage in collective action instead of private actions is key to the success of community-based (ecotourism) enterprises. This can only be realized when local communities are involved and get the support they need (Barnes et al., 2017; Barnes & van Laerhoven, 2015; Graham et al., 2019; Smit, n.d.).

However, concerning the understanding of CBET, there is still a mismatch between the knowledge provided in the literature and the knowledge required by society (Barnes & van Laerhoven, 2015). There is little evidence of the local communities knowing why their CBET enterprise failed or succeeded (Sakata & Prideaux, 2013). The initiatives are not learning enough from the success stories elsewhere and it is still uncertain which preconditions contribute to successful community-based ecotourism enterprises because of the dependency on local circumstances and their changes over time (Denman, 2001; Ngece, 2002; Romero-Brito et al., 2016). This mismatch in knowledge and ignorance of the preconditions is the knowledge gap for this research.

To highlight, it is necessary to explore what preconditions are important to address for the success of CBET enterprises. Therefore, to contribute to filling this knowledge gap and have a better understanding of the existing problems concerning CBET, exploratory research is conducted. The purpose of this research is to examine the preconditions for successful CBET by using a newly developed theoretical framework addressing the critical success factors (see section 2.2.1.) that were derived from previous studies about several concepts in the field of entrepreneurship, tourism, governance, and sustainability.

1.2. RESEARCH OBJECTIVE, AND MAIN QUESTION

The objective of this research is to provide explorative knowledge concerning critical success factors. This knowledge identified which factors are suitable as a precondition in the context of successful CBET. This contributes to the existing literature by providing insights from an enterprise's- and the community's perspective to decrease the knowledge gap described above.

This research aims (1) to create a set of preconditions for successful community-based ecotourism by analyzing the relative importance of the critical success factors and (2) to make recommendations based on the set of preconditions to endeavoring successful community-based ecotourism by applying the developed theoretical framework on a case study meta-analysis. Eventually, this set of precondition generates hypotheses for further research. Furthermore, (3) additional findings were found by the case study meta-analysis to make recommendations. These findings are related to the CBET definition, missing factors, and interlinkages for identifying configurations (clusters of CSFs). As guidelines through this research, the following research question (RQ) was used: *What factors contribute to the success of community-based ecotourism?*

To produce relevant insights to answer this question, this research analyzed a large-N of explicit CBET case studies to summarize the data via a quantitative case study meta-analysis which is seen as an efficient method to gain new insights that can provide scientific relevance for further research (Mengist et al., 2020). The new insights are gained by the theoretical framework that has been developed based on existing literature about various topics that are directly or indirectly related to community-

based ecotourism, e.g. collective action and conservation projects. This framework contains 19 critical success factors (CSFs), which are important to achieve successful community-based initiatives and were used to analyze the preconditions for successful CBET. The CSFs were checked if they indicate the success of CBET based on their 'relative weight' and 'direction' to the success of CBET. This was identified by looking at the frequency of CSFs, codes, and values explained in Chapter 3. Next, this framework was used in the case study meta-analysis which "can help to generate more precise estimates on the topic under study" by comparing and analyzing different case studies (Mengist et al., 2020, p. 2). Since the theoretical framework has never been applied, testing the framework on a broad set of case studies provided new insights into the relative importance of the CSFs.

1.3. RESEARCH FRAMEWORK AND SUB-QUESTIONS

To answer the main question and illustrate the broad lines of how the research objective was achieved, a research framework based on Verschuren & Doorewaard (2010) is shown in Figure 1.



Figure 1. Research framework. The boxes with bold lines are representing the focus of this research, namely applying the CSFs to the study meta-analysis to create the set of preconditions. The survey and semi-structured interviews are presented with dotted lines to emphasize that these results were used as additional methods for validation.

This research framework is a schematic representation of the five different steps that were taken to achieve the objective. Firstly, a literature review provided the input for the current state of the art (see Appendix I). This review addressed a broad set of CSFs from different topics that were found in Google Scholar and the UU library. Secondly, based on this review, a theoretical framework was developed. This list of factors has been narrowed down by looking at the frequency. The selected factors were then applied to different case studies in step 3 by conducting a case study meta-analysis. The case studies were derived from desk research and compared to each other to remove double cases and to exclude cases that were not suited for the analysis. Missing data were collected by asking questions that were

structured by the theoretical framework in semi-structured interviews and a survey. These two methods were applied to validate the case study meta-analysis and to provide additional information for the indepth analysis. Step 4 shows the results of the case study meta-analysis and in-depth analysis to provide the set of preconditions and eventually answer the RQ. Step 5 provides the discussion followed by the recommendations on how to endeavor community-based ecotourism, useful e.g. external actors and new initiatives, and ends with the conclusion.

The research framework was used to achieve the aim of this research. However, as guidelines through this research, a few sub-questions were used as steps to answer the main question and to make recommendations. These are as follows:

- (1) What are the critical success factors for CBET enterprises when looking at different concepts in the field of entrepreneurship, tourism, governance, and sustainability?
- (2) What are the scores of the critical success factors resulting from the case study meta-analysis?
- (3) What is the relative importance of the critical success factors looking at the 'relative weight'?
- (4) How is the relative importance influenced by the outcomes of the critical success factors looking at the 'relative direction'?
- (5) To what extent are the analyzed case studies addressing CBET, missing factors, and interlinkages between the CSFs?
- (6) What recommendations can be made based on the results?

1.4. SCIENTIFIC AND SOCIETAL RELEVANCE

The background and problem definition already highlight the knowledge gap indicating that further scientific research should analyze the success of CBET and what preconditions contribute to this. However, the preconditions are often explained for a specific case study (Jamaliah & Powell, 2018). For example, Parker & Khare (2005) attempted to create a framework for understanding the critical success factors for ecotourism in Southern Africa. But "there is limited research that focuses on evaluating the conditions needed to develop tourism systems that are resilient and adaptive to an uncertain future" (Jamaliah & Powell, 2018, p. 519). In earlier research, only a few researchers used a meta-analysis for identifying the success factors in the field of CBET (Fiorello & Bo, 2012; Musavengane & Simatele, 2017; Rhama, 2020; Romero-Brito et al., 2016; Stanley, 2014; Wondirad, 2019). But none of these studies present a set of preconditions for the success of CBET enterprises. A new theoretical body of knowledge on this topic can solve this problem which explains the reasoning for this exploratory research and why it is relevant in the field of science.

Next to this, CBET is an alternative approach for an equal distribution of tourism benefits and encompasses sustainable, environmental, cultural, and social components (Mtapuri & Giampiccoli, 2019). This approach contributes to the reduction of poverty, increase socio-economic development, and ensures nature conservation (Anup, 2017; Holden, 2010). Furthermore, several Sustainable Development Goals (SDGs) are addressed: the goals 'life on land' and 'sustainable cities and communities' (United Nations, 2020). However, an increase in awareness is necessary for the sustainable needs that conservation projects such as community-based ecotourism can promote (Heil, 2017). Therefore, this research is relevant to address social problems as well. By looking at the preconditions of successful community-based ecotourism, new insights can contribute to the actions for sustainability.

1.5. SCOPE OF THE RESEARCH

Community-based ecotourism is a complex phenomenon and is often analyzed in existing research. This approach addresses a wide orientation within the fields of entrepreneurship, sustainability, governance, and tourism. To define the theoretical framework, these fields were narrowed down into different concepts, namely collective action, conservation projects, community-based enterprises, and ecotourism, and resulted in the CSFs presented in the theoretical framework. These CSFs were analyzed within the case study meta-analysis to define the importance of these factors for the success of CBET. It was therefore not the goal to derive the CSFs from the case study-meta analysis as expected and already existing in the current literature, but to test the CSFs from the theoretical framework on the case study meta-analysis to determine the preconditions for successful CBET innovatively. The factors are thus not derived from the case studies. More importantly, the case study meta-analysis is the basis of indepth analysis and synthesis and therefore the main focus of this research.

The case study meta-analysis has been delineated to determine the bandwidth for the case studies since it is difficult to compare apples and oranges. Each case study has been investigated by a different researcher. Therefore, different approaches, research questions, methods, and CBET definitions were found in the case studies. To deal with these differences, this research applied criteria to the CBET definition to analyze this definition in a certain context and eliminate case studies that were not suitable for the analysis. Eventually, the case study selection ensured the delimitation which is explained in chapter three. This created the possibility to generalize the outcomes of the case study meta-analysis. However, standardization was not possible for the CBET definition. However, an attempt has been done and these results are used as an addition for making recommendations.

Lastly, community-based ecotourism includes a varying number and types of stakeholders such as non-governmental organizations, governmental authorities, local communities, tourists, tourism agencies, and project initiatives to name a few. For this research, the focus is on the CBET enterprises and their involved community, therefore the external actors such as the NGOs and government are excluded from the analysis. The focus is put on the local communities and their enterprise to provide a more detailed overview of preconditions within the given time frame of this research.

1.6. OUTLINE

This research is structured as follows. Firstly, chapter two describes the theoretical background explaining the important concepts and definitions, the CBET criteria, the critical success factors (CSFs), and the theoretical framework. Secondly, chapter three explains the case study meta-analysis, the case study selection, the coding process, in-depth analysis, and synthesis. Furthermore, this section explains which methods were used to obtain the data and these are the case study meta-analysis including desk research, semi-structured interviews, and survey. Thirdly, chapter four provides the results of the case study meta-analysis, the in-depth analysis, the synthesis, and the definition of CBET. Fourthly, chapter five presents the discussion wherein innovative aspects are highlighted and unexpected results, theoretical implications, and limitations of the research are discussed. Fifthly, in chapter six are the main and sub-questions answered in the conclusion. Lastly, in chapter seven, the recommendations for policy implications and further research are presented and are followed by the references and appendices.

2. THEORETICAL BACKGROUND

For this research, a theoretical framework was developed to analyze which of the CSFs are important to achieve successful CBET and eventually to provide the set of preconditions for successful CBET enterprises. This framework was developed based on existing literature addressing several CSFs related to collective action, conservation projects, community-based enterprises, and ecotourism. By combining these topics, a broader/comprehensive/holistic framework was developed to apply on sustainable issues regarding the success of CBET.

The following sections provide a better understanding of the scope of the research. In section 2.1., the concepts and definitions are described, including the CBET criteria. Section 2.2. explains how the theoretical framework was developed, including the CSFs. Eventually, the CSFs are visualized in the theoretical framework presented in section 2.3.

2.1. CONCEPTS AND DEFINITIONS

2.1.1. MAIN CONCEPTS AND DEFINITIONS

The research started with a literature review to derive the CSFs for the theoretical framework. There were four fields used as a starting point to obtain relevant background information, namely sustainability, entrepreneurship, governance, and tourism. These fields were narrowed down to four main topics from which the CSFs were derived (see Figure 2). Three topics are already explained in section 1.1., which are community-based enterprises, ecotourism, and collective action. The last topic, conservation projects are projects focusing on the sustainable use of natural resources and the protection of biodiversity, i.e. wildlife, endangered species, and critical places dealing with environmental destruction (National Geographic, 2019; World Wide Fund For Nature, n.d.).



Figure 2. Scope of literature review.

The key concepts for this research are CBET, the CSFs, and the set of preconditions. The CSFs are the independent variables contributing to the success of community-based enterprises. The dependent variable is the actual success of CBET enterprises. The set of preconditions included the CSFs contributing to the success of CBET enterprises. A distinction has been made between the focus of this research (i.e. set of preconditions) and further research (i.e. correlation between dependent and independent variables) since it was not possible to correlate the CSFs with the actual success of CBET enterprises within this time scope and due to the complexity of the case studies, e.g. no standardization possible. Furthermore, community-based ecotourism is entangled in the key concepts.

To elaborate on the concept of community-based ecotourism, it must be clear what ecotourism exactly is and is therefore defined by the following three core criteria: "(1) attractions should be predominantly nature-based; (2) visitor interactions with those attractions should be focused on learning or education; and (3) experience and product management should follow principles and practices associated with ecological, socio-cultural and economic sustainability" (Reimer & Walter, 2013, p. 122).

Furthermore, ecotourism contains several characteristics that also apply to community-based ecotourism: (a) travel to vulnerable and mostly protected areas; (b) strives to be low impact and small scale; (c) educates the tourist; (d) provides funds for nature conservation; (e) provides benefits for socioeconomic development and political empowerment; and (f) promotes to respect the norms and values from different cultures (Parker & Khare, 2005).

For this research, community-based enterprises and ecotourism are combined into the concept of CBET. The following general definition was used as a foundation to understand the concept of CBET. This definition was used due to the broad definitions described in the different case studies.

"A form of ecotourism where the local community has substantial control over, and involvement in, its development and management, and a major proportion of the benefits remain within the community. How the community is defined will depend on the social and institutional structures in the area concerned, but the definition implies some kind of collective responsibility and approval by representative bodies" (Denman, 2001).

2.1.2. COMMUNITY-BASED ECOTOURISM CRITERIA

Community-based ecotourism is often discussed in the existing literature. According to Reimer & Walter (2013) contains CBET four specific characteristics: "(a) principles of local participation, control or ownership of ecotourism initiatives; (b) a focus on environmental conservation and local livelihood benefits; (c) the promotion of customary and indigenous cultures; (d) the promotion of local and indigenous human rights and sovereignty over traditional territories and resources" (p. 123). However, it is still not obvious how to recognize community-based ecotourism in practice (Reimer & Walter, 2013). Therefore Reimer & Walter (2013, p. 125) found seven components as a result:

- (1) Involves travel to natural destinations;
- (2) Minimizes impact;
- (3) Builds environmental awareness;
- (4) Provides direct financial benefits for conservation;
- (5) Provides financial benefits and empowerment for local people;
- (6) Respects local culture; and
- (7) Supports human rights and democratic movements.

These components are justified in the case of Chi phat, in Cambodia, and therefore very specific which created the possibility that other case studies fell outside the boundaries of this research. Therefore, the components were revised into seven generalized criteria to analyze the definitions of CBET addressed in the different case studies (see Table 1).

| Table 1. List of general criteria for analyzing the definition of community-based ecotourism, | based on |
|---|----------|
| the components from Reimer & Walter (2013). | |

| Criteria | Description |
|------------------------|--|
| (1) Involves travel to | Involves travel to natural destination means the CBET enterprise is located |
| natural destinations | in a natural area under national/private, communal, or international |
| | protection and includes wilderness areas and ecotourism attractions. |
| (2) Minimizes | Minimizes impact refers to low environmental impacts and ecotourism |
| impacts | effects by the tourists, the tourism industry, and/or communities. Some |
| | examples are the amount of litter in the environment, the amount of erosion |
| | due to infrastructure, if tourists cut wood for own use, and if the |
| | accommodations use recycled local building materials and/or renewable |
| | energy. |
| (3) Builds | Builds environmental awareness is based on the opportunity to learn about |
| environmental | the people and place. This may ensure that local people and ecotourists |
| awareness | become more aware of their actions and impacts. The CBET enterprises |
| | should provide these opportunities by, for example, offering guides, |
| | workshops, internet sides providing extensive information about the area. |
| (4) Provides benefits | Provides benefits for conservation means that CBET contributes to the |
| for conservation | protection of nature. This is often linked to environmental awareness and |
| | financial benefits such as park entrance fees, funds, voluntary donations and |
| | conservation levies or indirectly gaining income for conservation from |
| | accommodations, restaurants, transport and guiding services. |
| (5) Provides benefits | Provides benefits (and empowerment) for local people is related to the |
| (and empowerment) | directly gained economic incentives from accommodations, restaurants, |
| tor tocal people | transport, and guiding services related to CBET as an alternative resource. |
| | Other benefits are provided in the form of empowering, opportunities for |
| | leadership skills and management skills, and market options to increase the |
| | tourism industry. |
| (6) Involves the local | Involves the local culture refers to the indigenous cultures addressed in |
| culture | CBET. For example, traditional dances or food workshops are offered as an |
| | touristic activity. Therefore, the cultural norms and values should be taken |
| | into account and respected by the tourists. |
| (7) Supports human | Supports human rights and land rights means that CBET takes the human |
| rights and fand rights | rights into account. Especially since the politics of CBET are seen as |
| | international development and the competition over land- and resource rights |
| | can result in conflicts. This can influence the role of ecotourism in a country. |

2.2. BUILDING A THEORETICAL FRAMEWORK

The concepts and topics are described in section 2.1.1. were used as keywords to search for critical success factors in the existing literature. To find such possible factors, a systematic literature review was used containing five steps, shown in Figure 3 (Khan et al., 2003). A complete overview of used sources is shown in the reference column in Table 2.



Figure 3. Steps to conduct a literature review.

To develop the theoretical framework, six steps were used in defining the CSFs and are as follows:

- (1) Search for claims and/or descriptions on critical success factors, failures, and challenges;
- (2) Define the most common critical success factor by counting the frequency of factors.;
- (3) Create the list of CSFs, which are expected to contribute to the success of CBET;
- (4) Narrow down this list by excluding the factors that were mentioned once;
- (5) Briefly explain the critical success factors (see the description in Table 2); and
- (6) Create a theoretical framework based on the final list of CSFs (see section 2.3).

For step 2, clustering has been carried out and is based on the article from Agrawal (2001). According to Agrawal (2001), many factors remain vague and unclear because of the generalization of terms and definitions. Existing work highlights the success factors of a specific topic, however, it is still uncommon to connect the different variables identified in causal chains. This indicates that models remain incomplete and conclusions are based on the specific studied sample, rather than applying the success factors more generally. Therefore, Agrawal (2001) suggested to cluster and classify the different variables. The structure of Agrawal's clustering is derived based on similar strategies from comprehensive studies by Elinor Ostrom, Robert Wade, and Jean-Marie Baland, and Jean-Phillipe Platteau, and he combined these studies for his cluster process.

Based on his clustering, Agrawal (2001) defined five categories (i.e. the first five in the enumeration), which are useful in creating a research design, case selection, and/or data collection. More importantly, the five categories were suitable to apply to the CBET enterprises, but they remain incomplete. Therefore, this research extended the clustering with one more category: support activities. This category 6 was added to analyze the kind of support external actors and the communities provide and are useful to gain insights on how to define the recommendations for, for example, external actors or new initiatives. To conclude, there are six categories used to identify the selected CSFs.

- (1) Resource system characteristics;
- (2) Group characteristics;
- (3) Relationship between resource system characteristics and group characteristics;
- (4) Institutional arrangements;
- (5) External environment including technology, market, and state characteristics; and
- (6) Support activities.

2.2.1. FINAL CLUSTERING

In total there were 19 critical success factors expected to be suitable for this research based on the six categories described above. Since the scope of the research is on the enterprises and their involved communities, the CSFs are only linked to the enterprise and community and not to other stakeholders such as NGOs, governmental authorities, etc. However, other stakeholders may be indirectly involved for example by funding or providing trainings.

Another aspect to note, the factors were analyzed separately and per-category but no clustering is defined between the CSFs. However, several interlinkages were recognized by analyzing the CSFs in the case studies. These interlinkages indicate that identifying configurations is a next step to look at in further research. This will increase the insights in necessary preconditions for success. The research from According to Baggio et al. (2016), identified configurations for successful common-pool resources to identify the combined positive effects of conditions and therefore a suitable guideline to follow. However, before configurations can be made, the preconditions must be identified and therefore the used scope in this research.

The CSFs were ordered into the six categories which are visible in Table 2. This list describes each factor to explain its meaning and understanding. Table 2 gives a brief overview of the selected factors and why these are relevant. More importantly, these factors are selected for defining the preconditions for successful community-based ecotourism and answers to sub-question one.

| Category | Critical success factor(s) | Description | Reference(s) | |
|--|---|--|---|--|
| 1. Resource system characteristics | InfrastructureAttractiveness of the environmentAlternative land use resources | <i>Resource system characteristics</i> refer to infrastructure such as transport, food, and water and accommodation facilities in the area. They refer to the attractiveness of the environment which is related to the willingness to pay by the tourists to go on holiday in the specific area or how satisfied the tourist are with their stay. Lastly, they refer to the alternative land use resources where the local people live on, e.g. agriculture, hunting, or natural resources. | Gruber et al. (2018); Hasan et al. (2020); Parker & Khare (2005) | |
| 2. Group | Appropriate leadership | <i>Group characteristics</i> refer to the existence of appropriate leadership in the | Barnes & van Laerhoven (2015); Dodds at al. (2018); Hamzah & | |
| characteristics | Wide awareness and knowledge of the initiative | how it's managed which means that the enterprise and community are aware | Mohamad (2012); Hasan et al. (2020); Manyara & Jones (2007); Ofori (2013); Runhaar & Polman (2018); Smit (n d.); Starling et al. | |
| | Local interests and motivation | of the activities and rules and understand them. Another factor is the local interests and motivation of the enterprise and community. Furthermore, the cultural and socio-economic differences should be in harmony, which means | | |
| | Cultural and socio-economic differences in harmony | that the norms and values of the enterprise and community are taken into account. | (2017); Thakadu (2005) | |
| 3. Relationship between 1. and 2. | Fair allocation of benefits | The relationship between resource system characteristics and group characteristics refers to the allocation of benefits which must be fair otherwise it will lead to conflicts. | Hasan et al. (2020); Parker & Khare (2005) | |
| 4. Institutional | Accountability | Institutional arrangements refer to the accountability of the project. Is the | Barnes & van Laerhoven (2015); Clarke (1999); De Koning et al. (2017); Hasan et al. (2020); Ofori | |
| arrangements | Communication throughout the project | project still in the hands of the national authorities or does the local community have responsibilities for actions and activities? They refer as | | |
| | Rule enforcement and appropriate rules-in-one | well to the communication throughout the project. For example, are there regular meetings to inform the community? Is its project mission and objectives clear for the enterprise and community? They refer as well to rule | (2013); Runhaar & Polman (2018); Smit (n.d.); Thakadu (2005); WONDIRAD at al. (2020) | |
| | Collaboration and participation in decision- making | enforcement and appropriate rules-in-one. This means that the rules are simple and easy to understand for everyone and that there is a good enforcement system. Furthermore, the community should collaborate and participate in decision-making to include their view and opinion. | wondikad et al. (2020) | |
| 5. External environment | Understanding of relevant state policies | <i>External environment</i> refers to technology, state, and market factors. The enterprise and community should understand the relevant state policies to | Barnes & van Laerhoven (2015); Clarke (1999); Gruber et al. (2018); | |
| | Monitoring and feedback | understand the rules, amendments, entitlements, and responsibilities | Hasan et al. (2020); Lee & Bond | |

Table 2. Final clustering of the CSFs per category, including their description and references.

| | Changes in biodiversity | concerning the management of CBET. Monitoring and evaluating the | (2018); Ofori (2013); Yalegama et |
|------------|--|---|--|
| | Using project plans as working document | project's outcomes over a certain period can give insights into the dilemmas and provide feedback. Especially long-term monitoring and local monitoring are necessary to provide accurate feedback. Changes in biodiversity refer to positive outcomes on biodiversity such as protecting species or less litter in the environment. Using project plans as a working document to update the enterprise and community. Especially because project requirements may change during the years and updates are regularly needed. | al. (2016) |
| 6. Support | Trainings | Support activities refer to training in, for example, leadership, hospitality, | Barnes & van Laerhoven (2015); |
| activities | Funding | and communication within the CBET industry. They refer to funding, e.g. | Catalano et al. (2019); Dodds et al. |
| | Autonomy | by the government or donors, to facilitate the enterprise and community to maintain the implemented projects in villages. Lastly, they refer to autonomy (self-governance) which is important to realize bottom-up project initiations and participatory planning. | (2018); Gruber et al. (2018); Hasan et al. (2020). Measham & Lumbasi (2013); Parker & Khare (2005); Smit (n.d.); Yalegama et al. (2016) |

2.3. THEORETICAL FRAMEWORK

The CSFs described in the section above are summarized in one theoretical framework (see Figure 4). By looking at these factors, new insights were gained for defining the preconditions for the success of CBET. The framework contains two boxes; critical success factors related to community-based ecotourism enterprises and critical success factors related to the support of external actors and communities. These boxes are connected by an arrow because the support activities provided by external actors or community (may) influence the success of CBET. Eventually, this framework is used as a guideline to find answers to the research question and sub-questions 2-6.





Figure 4. Theoretical framework illustrating the critical success factors of community-based ecotourism enterprises and external actors/communities per category. The striped arrow in the middle of the boxes means that the support activities (may) influence CBET.

3. RESEARCH METHODS

The purpose of this research is to examine the preconditions for the success of CBET enterprises. This chapter describes what research methods were used for collecting and analyzing the data, why these methods were used, which type of data was collected, how this data was collected, and which sources were used. Furthermore, this chapter explains how the concepts were measured and how the data were analyzed. To highlight, the reliability, suitability, and validity of the methods are addressed and the contribution to answering the RQ.

The chapter is divided into five sections structured following the mixed-method approach, indicating that both quantitative and qualitative methods were used. This is further elaborated in the sections. The common thread of the used methods is that it must examine the importance of the factors to identify the set of preconditions. Firstly, section 3.1. describes the case study meta-analysis which is the main focus of this research. Secondly, the selection of the case studies is presented in section 3.2. Thirdly, how the concepts were measured is explained by the operationalization in section 3.3. Fourthly, section 3.4 explains what data is collected and what data collection methods were used. Lastly, which analyses were used and how these were applied to the data is described in section 3.5.

3.1. CASE STUDY META-ANALYSIS

Often, case studies are used to illustrate the outcomes of community-based ecotourism in practice and provide in-depth information on a specific case (Verschuren & Doorewaard, 2010). By comparing a substantial set of case studies within a meta-analysis and applying the critical success factors on this meta-analysis, the theoretical framework was tested, learnings were derived from what is already done in the field, and new insights were gained (Ullah et al., 2017).

A quantitative meta-analysis was conducted to combine the results from multiple empirical studies and to summarize and analyze the case studies addressing CBET (Bryman, 2012; Mengist et al., 2020). This meta-analysis is a useful method to generate the data more accurately on a specific topic under study (Mengist et al., 2020). The generalization increased external validity (Bryman, 2012). It is reliable in dealing with a large dataset on secondary data where necessary information is missing or not certain enough (Baggio et al., 2016). Furthermore, it is already used in existing literature concerning CBET and therefore a suitable method for this research (Fiorello & Bo, 2012; Musavengane & Simatele, 2017; Rhama, 2020; Romero-Brito et al., 2016; Stanley, 2014; Wondirad, 2019).

To conduct this case study meta-analysis, several steps were followed which are based on the article of Mengist et al. (2020), wherein a literature review is combined with a quantitative meta-analysis and is therefore justified for this research. Firstly, the scope of the research was defined. In this case, the unit of analysis is community-based ecotourism. Secondly, suitable CSFs were defined and visualized in the theoretical framework. Thirdly, the strategy was defined. In this case, the comparative case study strategy was used to compare the studies to define the overarching characteristics (Yin, 2003). Fourthly, the case studies were found by carrying out desk research and selected based on the case study selection criteria. Fifthly, a synthesis was conducted in Excel to extract the data by using the coding steps to categorize the data and then using the coding process to score the data. Lastly, the scores were analyzed to extract meaningful information about CSFs. How the outcomes of the case meta-analysis were analyzed is described in section 3.5.1.

3.2. CASE STUDY SELECTION

For this research, a large-N of case studies were found increasing the external validity (Bryman, 2012). To highlight, the case study meta-analysis started with 56 articles, papers, and book chapters addressing single and comparative, quantitative, and qualitative cases. In total there were 87 different case studies discussed. Initially, these cases seemed suitable for the meta-analysis. But when looking at these cases in more detail, differences became visible. Therefore, the Case Study Selection Criteria were used to define the suitable cases (see section 3.2.1.) and a comparative case study strategy was used to identify the Case Study Characteristics (see section 3.2.2.).

3.2.1. CASE STUDY SELECTION CRITERIA

To narrow down the scope of the case studies and only include appropriate case studies, six case study selection (CSS) criteria were used (see Table 3). These criteria fit the research objective and help to find suitable case studies within the given amount of time for this research. However, there is a change that survival bias occurs which is a type of selection bias (Morgenstern, 2018). This means that only successful cases are found in the literature. Cases that are not successful are not valuable for research and often do not exist anymore due to their failures. This was taken into account when analyzing the case studies by CSS criteria 6. Furthermore, the CSS criteria ensured consistency when selecting the cases and therefore increased the reliability (Bryman, 2012).

Eventually, a total of 50 case studies were selected based on the six CSS criteria (see Table 3). These 50 case studies were derived from 33 articles, papers, and book chapters and excluded BSc- and MSc theses, and Ph.D. research, joint projects, and working papers to ensure quality empirical data. Furthermore, case studies that were described as developing stage or when the case studies is a challenging, potential, or promising initiative was excluded and articles, papers, or book chapters addressing the same CBET enterprise as already found emerged into one case study.

To highlight, the 50 case studies contained all six CCS criteria and were therefore used for the case-study meta-analysis to extend the existing theories concerning the success factors for CBET, to test the theoretical framework more precisely, and gain more insights for defining the preconditions. A detailed overview of the case studies is presented in Appendix II.

| Tabl | e 3. | Case | Study | Sel | ection | Critei | 1a. |
|------|------|------|-------|-----|--------|--------|-----|
| | | | | | | | |

| Crite | eria list |
|-------|--|
| 1. | The case study is described in an article, paper, or book chapter extracted from Google Scholar. |
| 2. | The topic addressed in the case study article, paper, or book chapter is explicit CBET. Other |
| | related topics such as tourism, ecotourism, natural resource management were not included. |
| 3. | The location of the case study is in a conservation area, national park, island, forest, mountain, |
| | or other kinds of a protected area. |
| 4. | The case study involves biodiversity protection and/or conservation. |
| 5. | The case study involves socio-economic development (e.g. poverty reduction). |
| 6. | The factors, conditions or other components analyzed in the case study lead to success. |

challenges, or failure of the CBET enterprise.

3.2.2. CASE STUDY CHARACTERISTICS

The comparative case study strategy from Yin (2003) was used to compare the studies to define the overarching characteristics. The main characteristics that are useful to understand the boundaries of the case studies emerged from comparing the similarities and differences of the cases and were found as overarching aspects. More importantly, these characteristics provide some general information and validate the outcomes of the case study meta-analysis since a broad perspective of case studies has been included. The found characteristics include: a specific form of case study, location of case study, distribution of locations, and involved stakeholders and provided the following information about the case studies.

All cases address CBET, however, some case studies describe the involved community while others mentioned different forms of enterprises. These are a community, village, project, network, homestay, sanctuary, eco-trek, lodge, camp, organization, company, association, site, and national park. No division was made between these forms and are included to make sure that the results apply to different kinds of cases.

Looking at the locations of these case studies, all cases are located in a conservation area under protection. However, a distinction was made between these locations whereby the areas are divided into (rain)forests, coastlines and wetlands, islands, game- and biosphere reserves, mountains, and (wildlife) national parks. Furthermore, the case study locations are distributed worldwide. Most of them are noteworthily located close to the equator in Central-America, Asia, and Africa but they are still widespread divided among the continents Asia, Africa, North-America, South-America, Europe, and Oceania (see Map 1).

The last characteristics address the involved stakeholders. Even though this research focused on the enterprise and community, the case studies itself presented different external actors as stakeholders that were a part of the CBET industry. Notable is that from the 50 case studies, 29 case studies addressed several types of external actors that contributed to the success of CBET by, for example, providing funds, giving trainings, or helping in the management. Most of these external actors are NGOs, governmental authorities, foundations and trusts, associations, tourism companies, and other projects.



Map 2. Overview of case study locations. A map is created with Google My Maps. Note that all 50 case studies are pinned on the map but not all are visible. The worldview display ensures that some location pins are overlapping when they are closely located to each other. Therefore, the map still shows the distribution of all case studies.

3.3. OPERATIONALIZATION

3.3.1. THE CRITICAL SUCCESS FACTORS

The CSFs were operationalized by looking at specific keywords. If a paragraph in the case study describes or mentions one of the keywords, it means that the paragraph explained some information about the related CSF. Table 4 summarizes which keywords were used for each CSF. By using these keywords for searching the CSFs in the case studies, a consistent structure is applied to the case studies which increases the reliability (Bryman, 2012).

| Category | Critical Success Factor | Keyword |
|-------------------|------------------------------------|---|
| 1. Resource | Infrastructure | Transport, airport, roads, park, |
| system | | accommodation, lodges, homestays, |
| characteristics | | restaurants, supermarket, provision |
| | Attractiveness of the environment | Willingness to pay, tourist's joy, tourist's |
| | | satisfaction, tourist's perception, tourist's |
| | | attitude |
| | Alternative land use resources | Natural resources, agriculture, hunting |
| 2. Group | Appropriate leadership | Leadership, chairman, chief |
| characteristics | Wide awareness and knowledge | Knowledge, learning, educate, inform, |
| | of the initiative | awareness |
| | Local interests and motivation | Interest, motivation, attitude, willingness |
| | | to participate, joy, satisfaction |
| | Cultural and socio-economic | Norms, values, discrimination, |
| | differences in harmony | competition, corruption, cultural impact, |
| | | social-economic impact |
| 3. Relationship | Fair allocation of benefits | Allocation of benefits, distribution of |
| between 1. and 2. | | benefits, sharing benefits |
| 4. Institutional | Accountability | Control, responsibility, external actors |
| arrangements | Communication throughout the | Meetings, consultation, communication |
| | Rule enforcement and appropriate | Bylaws enforcement rules regulations |
| | rules-in-one | Bylaws, enforcement, rules, regulations |
| | Collaboration and participation in | Decision-making |
| | decision-making | - |
| 5. External | Understanding of relevant state | Policies |
| environment | policies | |
| | Monitoring and feedback | Monitoring, feedback, evaluating |
| | Changes in biodiversity | Deforestation/reforestation, litter, |
| | | environmental management, human- |
| | | wildlife conflicts, biodiversity loss |
| | Using project plans as working | Project plan, CBET model, document, |
| | document | proposal |
| 6. Support | Trainings | Training, workshop, coaching |
| activities | Funding | Funding, financing |
| | Autonomy | Management, board, committee, |
| | | community involvement, community |
| | | participation, ownership |

Table 4. Operationalization.

3.3.2. CODING PROCESS

To analyze the CSFs in the case studies, a coding process has been used to score the case studies based on several codes and values in Microsoft Excel. This process contains a coding book as a starting point as described in the article by Baggio et al. (2016) and Cox et al. (2010). The underlying question regarding this codebook is if the case study addresses successful community-based ecotourism? Or not? Does it contain the CSFs? How important are these factors for the success of CBET? (Baggio et al., 2016).

This coding book contained a total of 19 variables which are the CSFs ordered into the six categories as explained in section 2.2. Each variable provided code as an outcome. The codes from the coding process defined by Cox et al. (2010) have been used as a basis. These codes were revised into the codes (1), (0), (-1), (X), and (Blank) used for scoring which is presented and explained in Table 5. Furthermore, each coded variable contains a value to indicate how much information was provided. There were three values used to indicate this: the value (+) when referred more than 6 times which means there is detailed information found, the value (+-) when referred between 2 - 5 times which means that there is little information found, and the value (-) when only referred once which means there is almost no information provided. Furthermore, during the coding process, the CBET definition been has analyzed based on the 7 different CBET criteria (see Table 1).

During the coding process, seven interpretative steps were used to search the CSFs in the case studies and to score these factors. Two extra steps were added to search for missing factors, possible interlinkages indicating configurations, and CBET definitions. These steps were followed by working with the program NVIVO and made the coding process replicable and reliable to analyze the variables. The steps are as follows:

- (1) Read the specific case study articles, i.e. textual database;
- (2) Orientate on the coding process including the possibility to switch between steps 3-4-5;
- (3) Start coding by labeling text fragments to the nodes i.e. the related CSFs;
- (4) Label new nodes if a new CSF is addressed in the case study to identify any missing factors in the theoretical framework;
- (5) Compare the nodes to search for the keywords which created eventually a trend of trade-offs used to code de variables more precise and in a transparent manner;
- (6) Note text fragments wherein a cluster of CSFs were described to identify possible interlinkages indicating configurations; and
- (7) End with selective coding to place the data in the related categories to eventually score the outcomes in Microsoft Excel.

| Variable | Code | Explanation |
|----------|-------|---|
| 19 CSFs | 1 | The case reported that the CSF exists with a positive effect on CBET. |
| | 0 | The case reported that the CSF does not exist with a negative effect on CBET. |
| | -1 | The case reported that the CSF exists with a negative effect on CBET. |
| | Х | The case reported conflicting evidence concerning the CSF if the text was |
| | | ambiguous, conflicting, or unclear. |
| | Blank | The case reported no outcome which means that the CSF was not addressed. |

Table 5. Overview of the used codes and their explanation.

3.4. DATA COLLECTION

Data is collected by using a triangulation of methods which increased the internal validity (Bryman, 2012). For this research, triangulation refers to "a process of cross-checking findings deriving from both quantitative and qualitative research" (Bryman, 2012, p. 392). The methods are described in the next sub-sections.

3.4.1. DESK RESEARCH

Desk research is used to search for existing case studies to gather facts and existing data to answer the RQ and the 2-6 sub-questions. In other words, no empirical data is gathered, but existing materials produced by others are used (Verschuren & Doorewaard, 2010). There are three characteristics involved: "(1) the use of existing material, in combination with reflection; (2) the absence of direct contact with the research object; (3) the material is used from a different perspective than at the time of its production" (Verschuren & Doorewaard, 2010, p. 194).

In this research, the use of existing materials consists of (peer-reviewed) articles, papers, and book chapters provided by scientists; and secondary data (e.g. survey and semi-structured interviews) compiled by the case study researchers. These materials are used within the case study meta-analysis. The absence of direct contact is related to the communities involved in CBET. This means that there is no data gathered during fieldwork. The material is used from a different perspective because it is used in a meta-analysis and a newly developed theoretical framework is applied to the case studies.

3.4.2. SURVEY

A survey is a quantitative data method and has been used to gain insights from (partly) successful CBET enterprises in practice (Verschuren & Doorewaard, 2010). The questions were related to the participants' views on the importance of each CSF and statements concerning the CBET definition and the CSFs. These questions were applied to their specific enterprise and involved community by using a Likert scale (totally agree – totally disagree for the statements and very important – not at all important for the importance). These insights were suitable to validate the case study meta-analyses outcomes and is used as an extra check on relative importance.

The survey was made via Survey Monkey and the participants were invited via mail. In total, 12 participants filled in the survey. The survey results are held unanimous because it is not important to gather information on the specific case study but more to validate the importance of the factors.

3.4.3. SEMI-STRUCTURED INTERVIEWS

Semi-structured interviews are a qualitative data method and have been used to gain more indepth/detailed information from the success of CBET enterprises in practice (Bryman, 2012). The survey questions were used as a guideline to identify the participant's view on the importance of each CSF and to statements concerning the CBET definition and the CSFs. Like the survey, these insights were suitable to validate the case study meta-analyses outcomes and are used as an extra check on the relative importance.

The participants were interviewed via phone call and online via Zoom, or email when there were connection problems or language barriers. In total, there were six participants interviewed and include the case studies presented in Table 6. These cases represent different forms of case studies and different locations and therefore seen as representatives. They are held unanimous because the answers were only used for validation.

| Cas | e study | Form | Location | Data interview |
|-----|--------------------------------|----------------------------|----------|-------------------|
| 1. | Tafi Atome Monkey Sanctuary | Community- | Ghana | July 10, 2020 |
| | (Referred to Tafi Atome) | based ecotourism project | | (via telephone) |
| 2. | Jaringan Ekowisata Desa | Village ecotourism network | Bali | June 13, and |
| | (Referred to JED) | | | July 20, 2020 |
| | | | | (via Zoom and |
| | | | | mail) |
| 3. | Toledo Ecotourism Association | Community-based | Belize | July 2, 2020 (via |
| | (Referred to TEA) | organization | | mail) |
| 4. | WILDTRUST | Community-owned game | South- | June 17, 2020 |
| | | reserve co-managed by | Africa | (via Zoom and |
| | | WILDTRUST | | telephone) |
| 5. | Casa Matsigenka Jungle Lodge | Community-based | Peru | June 23, 2020 |
| | (Referred to Casa Matsiguenka) | ecotourism lodge | | (via mail) |
| 6. | General information | General info about CBET | Belize | June 2, 2020 |
| | (Referred to as BELIZE) | in Belize | | (via telephone) |

Table 6. Overview of representative case studies participated in an online interview (or via email).

3.5. ANALYSIS

3.5.1. CASE STUDY META-ANALYSIS

The case study meta-analysis describes the quantitative generalizations as outcomes of the coding process and emphasized on the relative importance of the CSFs to summarize the key findings. The findings were summarized and compared by "conducting various analytical tests to show whether or not a particular variable affects the studies". (Bryman, 2012, p. 713). In this case, a coding technique and statistical description technique were used in Microsoft Excel.

The relative importance of the CSFs was determined based on three different components. Component 1 represents an overview of how much each CSF was found to give the estimated importance. This is described as the frequency of the CSFs giving 'relative weight' to the importance. The second component influences the importance and is related to the different codes used during the coding process to illustrate the positive or negative effects of the CSFs. The third component influences the importance as well and is related to the different values. These values explain how much information was found about the CSFs. Component 2 and 3 are described as frequency per code and frequency per value and indicate the 'direction' of the relative importance, i.e. positively or negatively influencing. The CBET definition was as well analyzed during the case study meta-analysis based on the definition used within the range of this research resulting in outcomes suitable for substantiating the recommendations.

To set validated thresholds, a descriptive statistical tool has been used to identify the spreading of the data points. This has been done by calculating the frequency distribution displayed in Boxplots in Microsoft Excel. The range is then based according to the values of min, max, interquartile (Q1), interquartile (Q2) = median, interquartile Q3, mean, and Interquartile range (IQR). By using these values, the outliers were defined indicting that the particular factor scored remarkably high on the addressed outcomes because of extreme values at the end of the distribution (Bryman, 2012). The boxplots are presented in the result section showing these outliers. The used data values are shown in Appendix III. The thresholds are presented next. Note that these are presented in frequency (number of

case studies). The relative frequency (outcome in %) is calculated by dividing the threshold's numbers by the 50 (= number total case studies).

The estimated importance is measured based on the number of case studies found. The higher the number, the more important the factor may be seen. To indicate this importance, the following importance scale has been used:

> Considered as important at a threshold of >35 case studies Considered as moderately important at 17 – 35 case studies Considered as not that important at a threshold of <17 case studies

The relative importance may be influenced by the outcomes of the codes which were identified by looking at the nuances presenting striking outcomes. Looking at these outcomes of the codes and values, they can positively influence the importance of the CSF: codes (1) and (0), and values (+) and (+-). Besides this, they can negatively influence the importance: code (-1) and value (-). Note that code (X) is excluded from the influence scale for identifying the influence since no clear information has been provided. However, this is presented as conflicting evidence. Code 0 is positively influencing because the lack of the CSF is resulting in a negative outcome on CBET which indicates that the existence of the CSF would be a positive outcome. The following thresholds were set to identify the striking outcomes.

For code (1)

Considered as positively influencing at a threshold of >22 case studies

Considered as moderately influencing at 8-22 case studies

Considered as not that much positively influencing at a threshold of <8 case studies

For code (0)

Considered as positively influencing at a threshold of >5 case studies

Considered as moderately influencing at 2-5 case studies

Considered as not that much positively influencing at a threshold of <2 case studies

For code (-1)

Considered as negatively influencing at a threshold of >3 case studies

Considered as moderately influencing at 1-3 case studies

Considered as not that much negatively influencing at a threshold of <1 case studies

For code (X)

Considered as conflicting at a threshold of >6 case studies Considered as a bit conflicting at 1 – 6 case studies Considered as not that much conflicting at a threshold of <1 case studies The following percentages highlight the thresholds for the striking outcomes:

Considered as a striking outcome at a threshold of > 44% for code 1 Considered as a striking outcome at a threshold of > 10% for code 0 Considered as a striking outcome at a threshold of > 6% for code -1 Considered as a striking outcome at a threshold of > 12% for code X

The relative importance may be influenced as well by the outcomes of the values which were identified by how much a CSF is referred to in NVIVO when looking at all 50 case studies. For this, the following thresholds were set to identify striking outcomes:

For value (+)

Considered as positively influencing at a threshold of >4 case studies Considered as moderately influencing at 1-4 case studies Considered as not that much positively influencing at a threshold of <1 case studies

For value (+-)

Considered as positively influencing at a threshold of >16 case studies Considered as moderately influencing at 6 – 16 case studies Considered as not that much positively influencing at a threshold of <6 case studies

For value (-)

Considered as negatively influencing at a threshold of >8 case studies Considered as negatively important at 4 – 8 case studies Considered as not that much negatively influencing at a threshold of <4 case studies

The following percentages highlight the thresholds for the striking outcomes:

Considered as a striking outcome at a threshold of >8% for value + Considered as a striking outcome at a threshold of >32% for value +-Considered as a striking outcome at a threshold of >16% for value -

3.5.2. IN-DEPTH ANALYSIS

The in-depth analysis continues on the results of the case study meta-analysis. In this analysis, the nuances are highlighted between the different categories, and the importance of the CSF is further discussed and validated by the responses obtained by the survey and semi-structured interviews. The relative importance was determined by the importance scale for component 1 which is based on the set thresholds (see the section above).

The influence scale of the codes and values continued from the set thresholds presented above and therefore these scales are not presented again as well. This scale can be interpreted as follows. If an outcome indicates positively influencing, then this is noted to strengthen the relative importance. If an outcome indicates negatively influencing, then this is noted as weaken the relative importance. Hereby means neutral that the relative importance is not strengthened or weakened and not that much means that there is little influence. The following percentages highlight the thresholds for the influence scale:

>44% for code (1) = positively influencing = strengthen the relative importance
>10% for code (0) = positively influencing = strengthen the relative importance
>6% for code (-1) = negatively influencing = weaken the relative importance
>8% for value (+) = positively influencing = strengthen the relative importance
>32% for value (+-) = positively influencing = strengthen the relative importance
>16% for value (-) = influencing negatively = weaken the relative importance

Lastly, the validity scale was used to identify the outcomes of the survey and interviews. These responses validated the relative importance of the CSFs. For the survey, the following scales were used to set the validation based on the Likert-scale (very important – not at all important). In this case, neutral means not validated, nor invalidated.

>9 participants answered important + very important = validated
3 - 9 participants answered important + very important = neutral
<3 participants answered important + very important = not validated

The validity scale for semi-structured interviews is as follows. In this case, neutral means not validated, nor invalidated as well.

>3 participants mentioned the factor positively or as important = validated 1 - 3 participants mentioned the factor positively or as important = neutral <1 participant did not mentioned the factor positively or not at all = not validated</p>

3.5.3. SYNTHESIS

The set of preconditions has been identified based on the outcomes from the case study meta-analysis, survey, and semi-structured interviews which are presented in the in-depth analysis in more detail. To identify the preconditions, the relative importance of the CSFs was determined and merged into a synthesis. This synthesis presents an overview of the estimated importance, the influence of the importance, and the validated outcomes concerning the importance of the CSFs.

To emerge the outcomes, the scales presented in Table 7 were used for filling in the synthesis Table. In this Table, the estimated importance is visualized by the symbols (+) for important, (+-) for moderately important, and (-) for not that important and validated by the (+) for validated, (+-) for remaining neutral (not validated, nor unvalidated) and (-) for not validated. To see if some factors were influenced, the synthesis Table is filled in with the following symbols: (+) indicates that the importance may be strengthened, (+-) indicates that the importance remained neutral (i.e. the estimated importance is not weakened or strengthened) and (-) indicates that the importance may be weakened (see Table 7). The total overview of relative importance (estimated + influence + validation) presents an indication of the preconditions for successful CBET.

| | Component 1 | Component 2 | Component 3 | Survey | Interview |
|-------------|-------------|-------------|-------------|---------------|---------------|
| | (Frequency) | (Codes) | (Values) | (Validity) | (Validity) |
| Used scale | Importance | Influence | Influence | Validity | Validity |
| | scale | scale | Scale | Scale | Scale |
| Symbols | (+/+-/-) | (+/+-/-) | (+/+-/-) | (+/+-/-) | (+/+-/-) |
| Explanation | Important/ | Strengthen/ | Strengthen/ | Validated/ | Validated/ |
| scale | Moderately | Neutral/ | Neutral/ | Neutral/ | Neutral/ |
| | important/ | Weaken | Weaken | Not validated | Not validated |
| | Not that | | | | |
| | important | | | | |
| | important | | | | |

Table 7. The importance, influence, and validity scales used for filling in the synthesis Table.

3.5.4. SET OF PRECONDITIONS

Eventually, the relative importance identified the set of preconditions. Because the relative importance is identified by both the 'relative weight' estimating the importance and the 'direction' influencing the importance, the order of the preconditions can be slightly different. Therefore, there ranking is based on three steps:

- (1) Rank the CSFs based on the estimated importance = list 1
- (2) Rank list 1 based on the validity (survey + interviews) = list 2
- (3) Rank list 2 based on the codes and values = list 3
- (4) List 3 represents the set of preconditions ranked by relative importance

4. RESULTS

The goal of this research was to create a set of preconditions for the success of CBET initiatives. This was done by applying the theoretical framework derived from the literature review on the case studymeta analysis. From this, several results were obtained and are presented in this section. These results address different insights concerning the relative importance and the effects of the CSFs, the different categories, and the definition of CBET.

To establish the final set of preconditions, two analyses have been conducted: a case study metaanalysis describing the outcomes of the coding process and determining the estimated importance (described in section 4.1.) and an in-depth analysis validating the outcomes of the case study metaanalysis per category, and determining the influence on the importance (described in section 4.2.). Eventually, a synthesis has evolved from these two analyses to determine the relative importance and provide the set of preconditions for the success of CBET (see section 4.3.).

4.1. CASE STUDY META-ANALYSIS

The case study meta-analysis emphasizes the relative importance of the CSFs, which is the main focus of this research. The relative importance of the CSFs was determined based on three different components. Per component, different thresholds were defined based on the distribution of the data to indicate the 'relative weight' of the CSFs (see section 4.1.1.). The first component represents an overview of how much each CSF was found to give the estimated importance (see section 4.1.2.). The second component influences the importance and is related to the different codes used during the coding process to illustrate the positive or negative effects of the CSFs. The outcomes of these codes are presented in section 4.1.3. The third component influences the importance as well and is related to the different values. These values explain how much information was found about the CSFs (see section 4.1.4.). From the case study meta-analysis were as well some additional findings found concerning several missing factors and possible interlinkages indicating configurations (see section 4.1.5). The CBET definition was as well analyzed during the case study meta-analysis and these results are presented in section 4.1.6.

During the case study meta-analysis, special attention was put to the generalized outcomes and that standardization was not possible. Therefore it is important to keep in mind that the presented results are pseudo precision. However, this precision is validated by the boundaries used for this research and is reliable due to the transparency and replicability of the coding process.

4.1.1. DISTRIBUTION PER COMPONENT

The defined thresholds were suitable for creating the scales with which the most important/interesting results and outliers were determined. More importantly, by using these thresholds, relative importance could be determined. Each component contains different randomized data and therefore a boxplot was suitable for looking at the frequency distribution to determine these thresholds. This is done by dividing the data into four parts of 25% (see method section 3.5.1.). The boxplots are visualized in Figure 5 to Figure 7. As shown, the boxplots recognized as well three outliers, one at code (-1) and two at value (+). The factors that are representing these outliers are described in the next sub-sections.



Figure 5. Boxplot component 1 - Frequency distribution of the CSFs. No outliers were found.



Figure 6. Boxplot component 2- Frequency distribution of the codes. The black dot indicates an outlier.



Figure 7. Boxplot component 3 - Frequency distribution of the values. The black dots indicate two outliers.

4.1.2. OVERVIEW OF THE FREQUENCY

How often the CSFs were found in the case studies (i.e. frequency) after finishing the coding process is visualized in Figure 8. The number given on top of each bar represents the number of case studies in which the specific factor was found. This number gives an estimation of the importance of each CSF. The higher the number, the more important the factor may be. Therefore, the bars in Figure 8 are arranged to ascend, which highlights the most important factors on the right and represents a range of possible factors that may play a role in the success of CBET. Furthermore, the Figure is displayed in different colors, one for each category, to represent how the different categories are divided.



Number of case studies wherein the CSF was found

Figure 8. Overview of frequency – Total number of case studies wherein a CSF was found. The six colors represent the six categories.

According to the estimated importance of the factors, five CSFs were found with a number above the threshold (>35 case studies) and are therefore considered as important for the success of CBET initiatives, namely *funding, infrastructure, alternative land use resources, autonomy,* and *accountability.* When looking at the threshold <17 five factors may be seen as not that important for CBET initiatives: *understanding of relevant state policies, using project plans as working document, rule enforcement and appropriate rules-in-one, appropriate leadership,* and *attractiveness of the environment.* The other nine factors with a number between 17-35 can be considered as moderately

important. These are monitoring and feedback, collaboration and participation in decision-making, wide awareness and knowledge of the initiative, local interests and motivation, cultural and socioeconomic differences in harmony, communication throughout the initiative, trainings, changes in biodiversity, and fair allocation of benefits.

When looking at the estimated importance per category, Figure 8 shows that category 6 – Support activities – scored higher on the importance scale (>35 case studies) with two of the three factors as important and one as moderately important with 32 case studies. Category 2 – Group characteristics – scored lower on the importance scale with three of the four factors seen as moderately important (between 17 and 35 case studies) and one factor as not that important (<17 case studies). This is as well applicable to category 5 – External environment – because this category contains two moderately important factors (24 and 18 case studies), and two factors that are not that important (3 and 10 case studies). Category 1 – Resource system characteristics – scored a bit divided on the importance scale with two of the three factors as important with 38 and 41 case studies but contains one factor as not that important with a score of 16 case studies. Category 4 – institutional arrangements – scored more divided because it contains one important factor (i.e. *accountability*) and contains two moderately important factors with 22 and 25 case studies and one factor not that important with 12 case studies. Category 3 – Relationship between category 1 + 2 – scored only moderately important however this category contains only one factor and therefore a division of factors is not possible.

Overall, three out of six categories contain several important factors (>35 case studies). Especially category 1 and 6 contain two factors above this threshold. Category 2 contains no factors that may be seen as important, but three out of four factors may be seen as moderately important. Category 5 contains two out of four factors as not that important. Furthermore, Category 4 is most divided when looking at the importance scale.

4.1.3. OVERVIEW OF THE CODES

The results from the coding process show that the relative importance is influenced by different code outcomes. The total number of case studies wherein a CSF was found has been split into four outcomes: the factor is seen as positive because of the existence of the factor, the factor is seen as negative because of the lack of the factor, the factor is seen as negative because of the existence of the existence of the factor and there is no outcome defined due to conflicting evidence. The CSFs outcomes were coded to indicate how often these outcomes were found. The overview of the number of case studies wherein the CSF was found per code looking at the 50 case studies (i.e. frequency per code) is shown in Figure 9.

Looking at this Figure, almost all factors scored highest on code (1) which means that the factors provide a positive outcome on CBET. Especially the factors *accountability, autonomy, alternative land use resources, infrastructure,* and *funding* scored above 22 case studies wherein the CFS exist with a positive effect. Notable, the factor *cultural and socio-economic differences in harmony* is the only factor with an equal number of case studies for the outcomes with code (1), (0), and (-1) and has a large amount of conflicting evidence compared to the other three codes. This result is as well visible when looking at Table 8 (relative frequency per code) where the results are presented in percentage to compare the CSFs outcomes by looking at the striking outcomes (see highlighted percentages in the Table) based on the influence scale.

According to this Table, almost all factors scored highest on code (1) (positive existing) indicating that almost all factors have a positive effect on CBET. There are a few percentages marked in green to highlight the highest percentages. *Infrastructure, alternative land use resources,*

accountability, funding, and *autonomy* all have a percentage above 44%. Besides, there are five factors worth mentioning that have the highest percentage of code (0) (non-existing negative) in contrary to the other factors. These are *infrastructure, cultural and socio-economic differences in harmony, fair allocation of benefits, accountability,* and *funding* which all scored above 10%.

However, there are as well some striking outcomes found indicating that some CSFs may influence CBET negatively and are therefore marked in red (see Table 8). When looking at the factors that are existing with negative effects (code (-1)), *autonomy* scored highest with 18% at a scale of >6% and is considered as an outlier, followed by *alternative land use resources*, and *cultural and socio-economic differences in harmony* both with 12% and *accountability* and *appropriate leadership* with 8%. The only factor that is noteworthy dealing with conflicting evidence when looking at the threshold of >12% is *changes in biodiversity*.

Looking at the different categories presented in Table 8, it is notable that two out of three factors in category 6 – Support activities – have a high percentage of code (1) (existing positive) but on the opposite one of these factors, namely *autonomy*, has a high percentage of code (-1) (existing negative). This applies as well to category 1 – Resource system characteristics – wherein the factor *alternative land use resources* has both a striking outcome on code (1) and (-1). Category 4 – Institutional arrangements – contains only one factor with striking outcomes, namely *accountability* scoring high on code (1), (0), and (-1). These three factors are therefore showing inconsistency. Furthermore, five out of six categories contain one or more factors that are marked in red (existing negative). Category 5 – External environment – contains no striking positive outcomes and is the only category with a high score on conflicting evidence. These results indicate that extra attention needs to be put on the differences within the categories, which can be found in section 4.2.1. whereas well attention is set on how much the code outcomes influence the importance.


Overview of the CSFs per code

Number of case studies wherein the CSF is existing with a positive effect (code 1)

Number of case studies wherein the CSF is non-existing with a negative effect (code 0)

■ Number of case studies wherein the CSF is existing with a negative effect (code -1)

Number of case studies wherein the CSF is existing with conflicting evidence (code X)



| Category | Critical Success Factor | or Codes (%) | | | |
|------------------------|--------------------------------------|-----------------------------|---------------------------------|------------------------------|--------------------------------|
| | | Positive existing (1) | Negative non-existing (0) | Negative existing (-1) | Conflicting evidence (X) |
| (1) Resource | Infrastructure | <u>50%</u> | <u>12%</u> | 2% | 12% |
| system | Attractiveness of the | 28% | 2% | 2% | 0% |
| characteristics | environment | | | | |
| | Alternative land use resources | <u>62%</u> | 4% | <u>12%</u> | 4% |
| (2) Group | Appropriate leadership | 14% | 6% | <u>8%</u> | 4% |
| characteristics | Wide awareness and | 30% | 10% | 0% | 6% |
| | knowledge of the initiative | | | | |
| | Local interests and | 32% | 10% | 2% | 2% |
| | motivation | | | | |
| | Cultural and socio- | 12% | 12% | 12% | 10% |
| | economic differences in harmony | | | | |
| (3) Relationship | Fair allocation of benefits | 34% | 16% | 6% | 12% |
| between category $1+2$ | | | | | |
| (4) Institutional | Accountability | 56% | 12% | <u>8%</u> | 10% |
| arrangements | Communication | 32% | 4% | 6% | 8% |
| C | throughout the initiative | | | | |
| | Rule enforcement and | 14% | 8% | 2% | 0% |
| | appropriate rules-in-one | | | | |
| | Collaboration and | 18% | 10% | 4% | 12% |
| | participation in decision- making | | | | |
| (5) External | Understanding of | 6% | 0% | 0% | 0% |
| environment | relevant state policies | | | | |
| | Monitoring and feedback | 22% | 6% | 0% | 8% |
| | Changes in biodiversity | 42% | 4% | 6% | <u>16%</u> |
| | Using project plans as | 12% | 8% | 0% | 0% |
| | working document | | | | |
| (6) Support | Trainings | 40% | 8% | 4% | 12% |
| activities | Funding | <u>46%</u> | <u>14%</u> | 2% | 12% |
| | Autonomy | <u>54%</u> | 4% | <u>18%</u> | 8% |

Table 8. Overview of the relative frequency (%) per code. The percentages representing striking outcomes are highlighted in bold underlined and color. The box represents the outlier.

4.1.4. OVERVIEW OF THE VALUES

The last component of the relative importance is related to the values that illustrate how much a CSF is referred to in NVIVO when looking at all 50 case studies (see Table 9). During the coding process were three values used: the value (+) when referred more than 6 times which means there is detailed information found, the value (+-) when referred between 2 - 5 times which means that there is little

information found, and the value (-) when only referred once which means there is almost no information provided. This indicates how much information is described in each factor. To link these values to the relative importance, it is assumed that the more information has been found, the more important the factor may be seen.

Table 9 shows the relative frequency (%) per value for the CSFs. The striking factors that are referred to more than six times indicate that the CSFs may positively influence the importance based on the threshold of >8%. These factors are *infrastructure, alternative land use resources, accountability,* and *autonomy*, whereby accountability and autonomy are considered to be outliers. The factors that can be seen as moderately influencing (>32%) are *infrastructure, alternative land use resources, fair allocation of benefits, accountability, changes in biodiversity, trainings, funding, and autonomy.* These three CSFs are referred to between 2-5 times. *Local interests and motivation, fair allocation of benefits, communication throughout the initiative,* and *funding* are negatively influencing the importance (>16%) since these have the highest percentage of 'referred ones'.

| Category | Critical Success Factor | Values (%) | | | |
|---------------------------|------------------------------------|-------------------|-----------------------|------------------|--|
| | | Referred >6 times | Referred 2-5 times | Referred once | |
| | | (+) | (+-) | (-) | |
| (1) Resource system | Infrastructure | <u>12%</u> | <u>42%</u> | 10% | |
| characteristics | Attractiveness of the environment | 2% | 20% | 10% | |
| | Alternative land use resources | <u>10%</u> | <u>52%</u> | 16% | |
| (2) Group characteristics | Appropriate leadership | 2% | 20% | 6% | |
| | Wide awareness and knowledge of | 8% | 18% | 14% | |
| | the initiative | | | | |
| | Local interests and motivation | 4% | 12% | <u>28%</u> | |
| | Cultural and socio-economic | 8% | 18% | 10% | |
| | differences in harmony | | | | |
| (3) Relationship | Fair allocation of benefits | 2% | 32% | <u>22%</u> | |
| between category $1+2$ | | | | | |
| (4) Institutional | Accountability | 28% | <u>34%</u> | 14% | |
| arrangements | Communication throughout the | 2% | 16% | <u>24%</u> | |
| | initiative | | | | |
| | Rule enforcement and appropriate | 4% | 12% | 8% | |
| | rules-in-one | | | | |
| | Collaboration and participation in | 4% | 12% | 16% | |
| | decision-making | | | | |
| (5) External | Understanding of relevant state | 0% | 0% | 6% | |
| environment | policies | | | | |
| | Monitoring and feedback | 0% | 14% | 14% | |
| | Changes in biodiversity | 6% | 32% | 14% | |
| | Using project plans as working | 0% | 8% | 12% | |
| | document | | | | |
| (6) Support activities | Trainings | 8% | <u>36%</u> | 8% | |
| | Funding | 4% | 30% | <u>28%</u> | |
| | Autonomy | 38% | 30% | 8% | |

Table 9. Overview of the relative frequency (%) per value. The bold underlined percentages represent the striking outcomes. The two boxes represent the outliers.

Looking at the different categories in Table 9, notable is that category 1 - Resource system characteristics – contains two out of three CSFs with a positive influence (i.e. *infrastructure* and *alternative land use resources*). The categories 4 - Institutional arrangements - and 6 - Support activities – contain only one striking factor that is positively influencing the importance. These two factors, *accountability*, and *autonomy* are considered as outliers as well. These two categories have as well one factor with 'referred once' as a striking outcome, namely *communication throughout the initiative* and *funding*. Besides this, category 2 contains no striking factors that are referring more than 6 times but only one factor scoring highest on value (-), namely *local interests and motivation*. This applies as well to category $3 - \text{Relationship between } 1 + 2 - \text{which has only one factor with 'referred once' as a striking outcome (i.e.$ *fair allocation of benefits* $). Category <math>5 - \text{External environment} - \text{has no striking outcomes at all when looking at these four factors.$

These results indicate as well that there are differences between the factors and within the categories. More in-depth information concerning the categories can be found in section 4.2.1. where attention is set on how much the value outcomes influence the importance as well.

4.1.5. ADDITIONAL FINDINGS

4.1.5.1. MISSING FACTORS

The case study meta-analysis focused on the 19 CSFs separately. However, there were other factors noticed in the 50 case studies that may contribute to the success of CBET. These other factors are therefore described as missing factors for this research, indicating that the theoretical framework is missing CSFs. Further research is needed to investigate this. The factors that were found above the threshold of 17 case studies were considered as important or moderately important. Table 10 summarizes the missing factors, indicating that these are all moderately important (between the threshold of 17 - 35 case studies). However, governmental support is indirectly related to *funding* and is therefore not completely missed as a factor.

| Missing factor | Description | Frequency |
|-------------------------|--|-----------|
| Marketing and promotion | Marketing and promotion is referred to as the marketing | 31 |
| | strategy, skills, and support, and promoting efforts for the | |
| | CBET industry. | |
| Governmental support | Governmental support is referred to as the support | 25 |
| | specifically provided by governmental authorities and | |
| | their involvement in CBET by e.g. funding, enforcement, | |
| | rules and regulations, and implementation of CBET. | |
| Gender (in)equality | Gender (in)equality is referred to as the inclusion and | 20 |
| | exclusion of women in the management, decision-making, | |
| | activities, jobs, or other aspects related to CBET. | |
| Market structure | Market structure is referred to as no competition between | 18 |
| | enterprises and communities, entrepreneurship, and | |
| | economic elements such as inflation, economic leakages, | |
| | costs and benefits, market opportunities, dictated prices, | |
| | and new forms of market-based livelihoods. | |

Table 10. Overview of missing factors, including their description and how often they were found in the case studies (frequency = number of case studies).

4.1.5.2. INTERLINKAGES BETWEEN THE FACTORS

During the coding process, several interlinkages were recognized between the CSFs and are summarized in Table 11 looking at the different case studies. These indicate that it is possible to identify configurations between the factors. Note that this list is not complete and no actual configurations were identified that may lead to successful CBET. However, by presenting these interlinkages as additional findings, it highlights the importance of identifying configuration to cluster the CSFs and see how the CSFs are arranged and work together. This will give a more detailed exploration of the importance of the factors for the success of CBET. Therefore, further research is needed to elaborate on this list and to identify the configurations.

Table 11. Noticed interlinkages between the CSFs useful for identifying configurations.

Interlinkages between CSFs

- Autonomy and accountability
- Autonomy, accountability, collaboration and participation in decision-making, and wide awareness and knowledge of the initiative
- Accountability, trainings, and funding
- Collaboration and participation in decision-making, communication throughout the initiative, and accountability
- Accountability and funding
- Changes in biodiversity, infrastructure, and alternative land use resources
- Alternative land use resources, infrastructure, and changes in biodiversity
- Collaboration and participation in decision-making and autonomy
- Gender inequality (i.e. missing factor), collaboration and participation in decision-making, and communication throughout the initiative
- Autonomy and leadership
- Accountability, funding, training, and marketing (i.e. missing factor)
- Funding, accountability, and changes in biodiversity
- Autonomy, local interests and motivation
- Autonomy, funding, governmental support and promotion

4.1.5.3. OVERVIEW OF CBET DEFINITION

The CSFs contribute to the success of initiatives such as CBET. To understand the concept of CBET more, this research analyzed the CBET definition as well. The reasoning for this is because the definition of community-based ecotourism is defined differently in the case studies. After all, each author used their perspective and definition of CBET in their research. This makes it complex to standardize the definition. However, to give a direction towards a more summarized and comprehensive definition, the seven criteria to recognize CBET were analyzed during the case study meta-analysis. This provided the following insights to understand the suitability of these criteria which is useful to make further recommendations to attempt standardizing the conceptualization of CBET.

Table 12 shows that only three criteria were found in all 50 case studies, namely (1) involves travel to natural destinations, (4) provides benefits for conservation, and (5) provides benefits (and empowerment) for the local people. This emphasizes that these criteria are involved in all the definitions addressed in the case studies and therefore suitable elements for recognizing CBET. The criteria (2) minimizes impacts, (3) builds environmental awareness, and (6) involves the local culture are often

found – between 32 and 40 times – and therefore also seen as well suited elements. The last criteria, (7) *support human rights and land rights*, is only found eight times in all the case studies which may indicate that this element for recognizing CBET is not that suitable.

| Criteria | Frequency |
|--|-----------|
| (1) Involves travel to natural destinations | 50 |
| (2) Minimizes impacts | 38 |
| (3) Builds environmental awareness | 32 |
| (4) Provides benefits for conservation | 50 |
| (5) Provides benefits (and empowerment) for the local people | 50 |
| (6) Involves the local culture | 40 |
| (7) Supports human rights and land rights | 8 |

4.2. IN-DEPTH ANALYSIS

The results described in this section follow from the outcomes of the case study meta-analysis described above in section 4.1. The in-depth analysis zooms in on the six different categories used within the theoretical framework by looking at the case study meta-analysis and the responses obtained from the survey and interviews. Therefore, this analysis is structured by the different categories to emphasize the nuances (see section 4.2.1.). This section presents the most valuable insights concerning the relative importance gained for this research by looking at the three components. Furthermore, insights concerning the CBET definition were obtained from the interviews and survey responses as well, which are presented in sections 4.2.2.

4.2.1. THE OUTCOMES PER CATEGORY

The following sub-sections describe the outcomes of the case study-meta analysis per category and zoom in on relative importance. These outcomes of the frequency and codes are visualized in Figure 10 to Figure 15. The outcomes of the values are visible in the overview Table 9 and are here described per category. The Figures show for each category the total number of case studies wherein the CSF was found (i.e. frequency) and how often they were found per code (i.e. frequency per code).

The relative importance of CSFs is influenced by different codes. This can be positively influenced (code (1) and (0)), negatively influenced (code (-1)) or no influence defined (code X) which indicates the effect of the factor (i.e. the factor is positively (code (1) and (0)) or negatively (code -1) affecting CBET). The relative importance is influenced by the different values as well. This can be positively influenced (value (+) and (+-)) and negatively influenced (value (-)). Furthermore, the interview and survey responses provide additional information concerning the CSFs to check the outcomes on validity. Some differences in responses were recognized and are described per category.

4.2.1.1. CATEGORY 1: RESOURCE SYSTEM CHARACTERISTICS

This category contains three CSFs: *infrastructure, attractiveness of the environment,* and *alternative land use resources.* Looking at the estimated importance, two of these, *infrastructure* and *alternative*

land use resources scored high (> 35 case studies) on the total number of the case studies the CSF was found in and are therefore considered as an important factor (see Figure 10).

Looking at the codes (see Figure 10), only *infrastructure* and *alternative land use resources* scored above the threshold of 22 case studies on code (1). Next to this, *infrastructure* scored as well above the threshold of code (0) (>5 case studies). This indicates that these two factors influence positively the importance. However, *alternative land use resources* scored as well above the threshold on code (-1) (>3 case studies), indicating that this factor may as well negatively influences the importance. The factor *attractiveness of the environment* scored moderately influencing with 14 case studies on code (1).

Looking at the values, the factors *infrastructure* and *alternative land use resources* scored both >8% for the value (+) and above 32% for the value (+-) and are therefore positively influencing the importance. The *factor attractiveness of the environment* scored moderately on all three values.

Box 1. Conclusion Case Study Meta-analysis – Category 1.

To conclude, two CSFs are estimated as important: *infrastructure* and *alternative land use resources*. Only the importance of *infrastructure* can be strengthened based on the codes and values. The importance of *attractiveness of the environment* and *alternative land use resources* have remained neutral (i.e. the importance is not strengthened or weakened) because of inconsistency or moderately scores.





Infrastructure Attractiveness of the environment Alternative land use resources

Figure 10. Outcomes of case study meta-analysis – Frequency and frequency per code – Category 1.

Regarding the survey, two scored the factor *infrastructure* as neutral, six scored the factor as important, and four as very important. *Attractiveness of the environment* scored eight times as very important and four times as important. This in contradiction to the estimated importance defined in the case study-meta-analysis. Therefore, the relative importance of these two factors is validated. *Alternative land use resources* scored inconsistently because two scored the factor as very important, six as important, two as neutral and two as not that important. Therefore, the relative importance of this factor remained neutral (i.e. not validated, nor invalidated).

Regarding the interviews, all three factors were mentioned in the interviews. However, the factor *infrastructure* was mentioned with contradicting information. According to JED, creating awareness is more important than infrastructure to work as a collective action together. Then existing infrastructure can be used, such as empty spaces for homestays, instead of building new accommodations (personal communication, June 13, 2020). According to WILDTRUST, a good infrastructure is very important to ensure access and good roads for tourists (personal communication, June 17, 2020). The factor *attractiveness of the environment* was only mentioned in one interview as a positive aspect. Therefore, the relative importance of these two factors remained neutral (i.e. not validated, nor invalidated). The factor *alternative land use resources* is mentioned positively in four of the six interviews and therefore validated as relative importance for the success of CBET. Farming and hunting were often found in the case studies. This is confirmed by JED indicating that the rice fields are a part of their culture and income (personal communication, June 13, 2020) and by WILDTRUST highlighting that hunting is important to supply meat to the community in terms of hunger and poverty (personal communication, June 17, 2020).

4.2.1.2. CATEGORY 2: GROUP CHARACTERISTICS

This category – Group characteristics – contains four CSFs: *appropriate leadership, wide awareness and knowledge of the initiative, local interests and motivation,* and *cultural and socio-economic differences in harmony.* Looking at the estimated importance, none of these factors can be considered as important since they all scored <35 case studies. However, only *appropriate leadership* scored <17 and is therefore considered as not that important. The other 3 factors scored between 17 – 35 case studies and are therefore considered as moderately important (see Figure 11).

Regarding the codes, *wide awareness and knowledge of the initiative* and *local interests* and motivation scored between the threshold 8 - 22 case studies for code (1) indicating they are moderately influencing the importance and the other two factors have not that much influence. The factor *cultural and socio-economic differences* scored high on code (0) with a score of 6 (>5 for influencing the importance). However, this factor scored as well high with a score of 6 on code -1, which is above the threshold of 3 for code (-1), indicating that this factor may influences the importance negatively and therefore shows inconsistency. The factor *appropriate leadership* scored as well above this threshold for code (-1) and is therefore considered as negatively influencing the importance (see Figure 11).

Regarding the values, all factors scored moderately on the influence scale for value (+) and value (+-). Only the factor *local interests and motivation* scored above the threshold of 8 (16%) for the value (-) considering this factor as negatively influencing the importance.

Box 2. Conclusion Case Study Meta-analysis – Category 2.

To conclude, none of these CSF are estimated as important and only *appropriate leadership* is estimated as not that important. None of these factors are strengthen the importance based on the codes and values. Only the importance of *appropriate leadership* can be weakened based on the codes and the importance of *local interests and motivation* can be weakened based on the values. All the other factors have remained neutral and therefore their importance is not strengthened or weakened because of inconsistency or moderately scores.



Category 2. Group characteristics

Regarding the survey, all factors scored high on the validity scale. *Appropriate leadership* scored 9x on very important and 3x on important. This is in contradiction to the estimated importance defined in the case study meta-analysis. The factor *wide awareness and knowledge of the initiative* scored 10x very important and 2x important, the factor *local interests and motivation* scored 11x very important and 1x important, and the factor *cultural and socio-economic differences in harmony* scored 7x very important 4x important and 1x neutral. Therefore, the relative importance of these four factors is validated.

Regarding the interviews, all four factors were mentioned in the interviews. However, the factor *appropriate leadership* was mentioned with contradicting information. For example, at the association TEA, the leaders are trained and educated to "adhere to the policies, rules, and regulations of the Association" (personal communication, July 2, 2020). This is in contrast with leaders that are mistrusted and seen as a problem (Tafi Atome, personal communication, July 10, 2020). The factors *local interests and motivation* and *cultural and socio-economic differences in harmony* are only mentioned positively in one interview. Therefore, the relative importance of these factors remained neutral. The relative importance of *awareness and knowledge of the initiative* is validated because it is mentioned in four of the six interviews as a positive aspect. To highlight, this factor is part of the mission of JED "which is to build awareness, confidence, and pride of local communities of their own culture" (personal communication, June 13, 2020).

4.2.1.3. CATEGORY 3: RELATIONSHIP BETWEEN CATEGORY 1 + 2

This category only contains one CSF which is *fair allocation of benefits* and is estimated as moderately important since this factor scored 34 case studies, which is between the threshold 17 - 35 case studies (see Figure 12). Regarding the codes, this factor scored only above the threshold of 6 on code (0) indicating that this factor is positively influencing the importance. The CSF is moderately influencing the importance when looking at the other codes (see Figure 12). Regarding the values, this factor scored

Figure 11. Outcomes of case study meta-analysis – Frequency and codes – Category 2.

above the threshold of 8 for the value (-) indicating that this factor is negatively influencing the importance.

Box 3. Conclusion Case Study Meta-analysis – Category 3.

To conclude, the CSF fair allocation of benefit is estimated as moderately important. The importance of this factor can be strengthened based on the codes and can be weakened based on values.



Figure 12. Outcomes of case study meta-analysis – Frequency and frequency per code – Category 3.

Regarding the survey, the factor fair allocation of benefits scored 1x neutral, 4x important, and 7x very important. Therefore, the relative importance of this factor is validated.

Regarding the interviews, *fair allocation of benefits* is mentioned positively in five out of six interviews and therefore validated as an important factor. According to Casa Matsiguenka, a rotating system is existing to create employment among members. This provides income to share with their own families, however, this is still challenging since the collective benefits are insufficient (personal communication, June 23, 2020). In the conversation with JED is said that the revenues are for the communities. Specifically, 85% of the income goes to the village and 15% of the income goes to JED itself for promoting and marketing and providing activities and trainings in cooperation with Wisnu Foundation (personal communication, June 13, 2020).

4.2.1.4. CATEGORY 4: INSTITUTIONAL ARRANGEMENTS

The category – Institutional arrangements – contains the CSFs accountability, communication throughout the initiative, rule enforcement and appropriate rules-in-one, and collaboration and participation in decision-making (see Figure 13). When looking at the importance scale, the factor accountability scored above the threshold of 35 case studies and is therefore considered as important. The factor rule enforcement and rules-in-one scored <17 case studies and is therefore considered as not that important. The other two factors are considered as moderately important.

Looking at the codes, the factor *accountability* scored with 28 case studies above the threshold of 22 case studies for code (1) and with 6 case studies above the threshold of 5 case studies for code (0) indicating that the importance is influenced positively. However, this factor scored as well with 4 case studies above the threshold of 3 case studies for code (-1) indicating that the importance is influenced negatively. Therefore, this factor shows inconsistency. The other three factors scored moderately influencing the importance and not that much influencing the importance based on the codes.

Looking at the values, the factor *accountability* scored with 28% high above the threshold of 8% for value (+) indicating that this outcome is an outlier and therefore extremely noteworthy. The factors scored as well above the threshold of 32% for value (+-). Therefore, the importance of this factor is positively influenced based on these values. The factor *communication throughout the initiative* scored with 24% above the threshold of 16% for value (-) indicating that the importance of this factor is influenced negatively based on this value. The other two factors scored neutral indicating that the importance is moderately influenced.

Box 4. Conclusion Case Study Meta-analysis – Category 4.

To conclude, the CSF *accountability* is estimated as important, the CSFs *rule enforcement and appropriate rules-in-one* as not that important and the other two CSFs as moderately important. Based on the codes, the importance of *accountability* remained neutral due to inconsistency. The importance of the other three factors remained neutral due to moderately outcomes. The factors are therefore not strengthening or weakening the importance. Based on the values, the importance of *accountability* can be strengthened especially since this is a positive outlier. The importance of *communication throughout the initiative* can be weakened because this factor is often only referred once.





Figure 13. Outcomes of case study meta-analysis – Frequency and frequency per code – Category 4.

Regarding the survey, all factors scored high on the validity scale. The factor *accountability* scored 8x very important and 4x important. Note that *accountability* is referred to as the responsibility to the community in the survey. The factor *communication throughout the initiative* scored 8x very important, 3x important, and 1x no answer filled in. *Rule enforcement and appropriate* rules-in-one

scored 2x very important and 10x important. This is in contradiction to the estimated importance defined in the case study meta-analysis. Lastly, collaboration and participation in decision-making scored 8x very important and 4x important. These results indicate that the relative importance of all four CSFs is validated.

Regarding the interviews, the factors accountability and communication throughout the initiative are validated as important factors because these factors are mentioned positively in five out of six interviews. Accountability is often referred to as an external actor in the case studies. This is as well confirmed in the interviews. For example, "One of JED founders is an environmental NGO called Wisnu Foundation". JED is working together with this foundation because this NGO "works with other villages who are interested in adapting JED's eco-tourism concept and principles" (personal communication, July 20, 2020). In the Somkhanda Game Reserve is the NGO WILDTRUST co-managing with the Gumbi community in the form as an exit strategy. The goal is to give it in hands of the community within 5 years (personal communication, June 17, 2020). In Belize is the Belize Tourism Board (BTB), Belize Industry Association, and the Belize Hotel Association accountable for CBET in general (personal communication, June 2, 2020). However, "The BTB is mandated to promote Belize as a tourist destination but they are just promoting luxurious hotels, the caves, and the beaches" and excluding TEA. The success of TEA could be increased if they were included by the BTB and supported by the government and other locally Belizean NGOs for funding, monitoring, and evaluation (personal communication, July 2, 2020). The relative importance of rule enforcement and appropriate rules-inone is as well-validated because this factor is mentioned in four out of six interviews. The factor collaboration and participation in decision-making is only mentioned in two interviews and therefore the relative importance remained neutral.

4.2.1.5. CATEGORY 5: EXTERNAL ENVIRONMENT

The category External environment contains the following four CSFs: *understanding of relevant state policies, monitoring and feedback, changes in biodiversity,* and *using project plans as working document* (see Figure 14). Two of these factors are estimated as moderately important with a score between 17 - 35 case studies, namely *monitoring and feedback* and *changes in biodiversity*. The other two factors scored <17 case studies and are therefore considered as not that important.

Regarding the codes, all factors are moderately influencing the importance based on the codes (1), (0), and (-1). Notable is that the factor *changes in biodiversity* is the only factor that scored high on conflicting evidence. The factor scored with 8 case studies above the threshold of 6 case studies for code (X) indicating that this factor needs more information to conclude the influence on the importance. Regarding the values, all factors are moderately influencing the importance based on the values and therefore remained neutral.

Box 5. Conclusion Case Study Meta-analysis – Category 5.

To conclude, none of these CSFs are estimated as important. Two CSFs, *monitoring and feedback* and *changes in biodiversity* are estimated as moderately important and the other two as not that important. Based on the codes and values, the importance of all factors remained neutral due to moderately outcomes. Only the factor *changes in biodiversity* scored high on code (X) indicating conflicting evidence. The factors are therefore not strengthening or weakening the importance.



Category 5. External environment

Figure 14. Outcomes of case study meta-analysis – Frequency and frequency per code – Category 5.

Regarding the survey, *understanding of relevant state policies* scored five times as very important, six times as important, and once as low importance. This indicates inconsistency and therefore the relative importance has remained neutral. The factor *monitoring and feedback* scored eight times as very important and four times as important. *Changes in biodiversity* scored five times as very important, six times as important, and one time as neutral. Lastly, the factor *using project plans as working document* scored three times as very important and nine times as important. Therefore, the importance of these factors are validated.

Regarding the interviews, only the factor *changes in biodiversity* is validated as an important factor because this factor is mentioned positively in four interviews. For example, the Tafi Atome Monkey Sanctuary has been able to protect and conserve the forest (about 100 ha), and protect the monkeys and other animals in the forest (personal communication, July 10, 2020). And TEA is successful "because we still maintain, control, own, and use our natural resources sustainably by helping it as a result of the many educational workshops on environmental conservation" (personal communication, July 2, 2020). The relative importance of monitoring and feedback remained neutral. The other two factors, *understanding of relevant state policies* and *using project plans as working document*, are the only factors that are not mentioned in the interviews and therefore their relative importance is not validated.

4.2.1.6. CATEGORY 6: SUPPORT ACTIVITIES

The last category contains the CSFs *trainings, funding,* and *autonomy*. The factors *funding* and *autonomy* are estimated as important with a score of >35 case studies and *trainings* as moderately important with a score between 17 - 35 case studies (see Figure 15).

Looking at the codes, the factor *funding* scored above the threshold of 22 case studies for code (1) and above the threshold of 5 case studies for code (0) and is therefore considered as positively influencing the importance. The factor *autonomy* scored as well above the threshold of 22 case studies

for code (1) indicating that the importance is influenced positively. However, this factor scored with 9 case studies above the threshold of 3 case studies for code (-1), highlighting that this score is considered as a negative outlier. Therefore, this factor shows inconsistency. The factor trainings is moderately influencing the importance based on the codes.

Looking at the values, the factor *autonomy* scored with 38% high above the threshold of 8% for value (+) indicating that this outcome is an outlier and therefore extremely noteworthy. The factor *trainings* scored with 36% above the threshold of 32% for value (+-). Therefore, these two factors are positively influencing the importance based on these values. The importance of *funding* is negatively influenced based on value (-) since this factor scored with 28% above the threshold of 16%.

Box 6. Conclusion Case Study Meta-analysis - Category 6.

To conclude, the CSFs *autonomy* and *funding* is estimated as important and the CSF *trainings* as moderately important. Based on the codes, the importance of *autonomy* remained neutral due to inconsistency (i.e. the importance is not strengthened or weakened). Furthermore, the importance of *funding* can be strengthened and the importance of trainings remained neutral. Based on the values, the importance of *autonomy* and *trainings* can be strengthened. The importance of *funding* can be weakened because this factor is often only referred once. Notable is that the factor *autonomy* contains two outliers, one positive for value (+) and one negative for code (-1) which emphasizes the need for more in-depth information.



Figure 15. Outcomes of case study meta-analysis – Frequency and frequency per code – Category 6.

Regarding the survey, all factors scored high on the validity scale. *Trainings* scored 6x very important and 6x important. *Funding* scored as well as 6x very important and 6x important. *Autonomy* scored 8x very important and 4x important. These results indicate that the relative importance of all these CSFs is validated.

Regarding the interviews, the relative importance of all three factors is validated because the factors *funding* and *trainings* are mentioned positively in four interviews. According to WILDTRUST,

"Funding is extremely important to become self-funding eventually" (personal communication, June 17, 2020). According to TEA, CBET will be less successful if financial support disappears from e.g. governmental authority. Furthermore, TEA has leadership trainings, trainings arranged by Toledo Institute for Development and the Environment (TIDE), and trainings organized by the District Executives for refreshing the members at the beginning of the tourist season (personal communication, July 2, 2020). The factor *autonomy* is mentioned positively in all six interviews. However, autonomy is for each particular case study slightly different. For example, JED is a village ecotourism network wherein the communities "working as a collective action together". The structure consists of three levels: Wisnu Foundation as founders and community owners, one Trustee to evaluate JED's performance and give advice, and the management level containing a manager, administration and accounting staff, and the program coordinator (personal communication, July 20, 2020). Casa Matsiguenka belongs to two native communities with two managers (one from each community) for the lodge (personal communication, June 23, 2020). WILDTRUST is an NGO that is co-managing with the community. However, the game reserved is owned by the Emvokweni Community Trust (ECT) representing the Gumbi community (personal communication, June 17, 2020). And TEA is a community-based program, democratically electing or re-electing their leaders every 3 years. This program has two levels of governance, one District Executive Committee (oversees the management of the office) and one Village Group Executives (coordinates and managed the guesthouse and tours in their community) (personal communication, July 2, 2020).

4.2.2. CBET DEFINITION IN PRACTICE

The survey also addressed the extent to which the participants agreed with the statements regarding the CBET criteria. These are the same criteria that were defined as suitable for recognizing CBET. The responses are summarized in Table 13.

| Criteria | Frequency | | | | | |
|---|-----------|----------|---------|-------|----------|------------|
| | Strongly | Disagree | Neutral | Agree | Strongly | Not |
| | disagree | | | | agree | applicable |
| (1) Involves travel to natural destinations | 0 | 1 | 1 | 3 | 7 | 0 |
| (2) Minimizes impacts | 0 | 0 | 0 | 5 | 6 | 1 |
| (3) Builds environmental awareness | 1 | 0 | 1 | 3 | 7 | 0 |
| (4) Provides benefits for conservation | 0 | 1 | 1 | 5 | 5 | 0 |
| (5) Provides benefits (and empowerment) for the | 0 | 0 | 2 | 3 | 7 | 0 |
| local people | | | | | | |
| (6) Involves the local culture | 0 | 0 | 0 | 5 | 7 | 0 |
| (7) Supports human rights and land rights | 0 | 0 | 1 | 5 | 5 | 1 |

Table 13. Overview of survey responses regarding the criteria for CBET. The frequency indicates how many participants agreed (based on the Likert scale) with the particular criteria for their enterprises.

As shown, most of the 12 participants (10 or more) agreed or strongly agreed that the particular criteria apply to their enterprise and involved community. This indicates that the criteria are suitable for recognizing CBET in practice and not only from a theoretical perspective.

These criteria are all contributing to the goal of CBET and therefore useful for enterprises and communities to take into account when figuring out which aspects of CBET can be developed, improved, or sustained since CBET itself is a broad definition. As Table 13 shows, environmental awareness is once overlooked, involves travel to natural destinations once, and provides benefits for conservation is overlooked in one particular case. This may indicate that not all criteria of CBET are addressed in the case studies while these criteria represent important aspects of CBET.

4.3. SYNTHESIS

The relative importance of the CSFs is defined by the three components and validity and their outcomes are presented in a synthesis showing the differences in outcomes. A summarized overview of the outcomes per category is presented in Table 14. By comparing these CSFs outcomes, the relative importance was determined. The outcomes of component 1 indicated that five factors are estimated as important and five as not that important. The moderately important factors are considered as important as well since the CSFs represent the expected factors for the success of CBET.

The factors were checked on validity by the survey and interview responses. Looking at the total overview of the survey responses, almost all factors scored high on the importance scale which validates that these CSFs are important for the success of CBET. Only the factor *alternative land use resources* and *understanding of relevant state policies* scored inconsistently and are therefore remained neutral (not validated, nor invalidated). Looking at the total overview of interview responses, the only *project plans as working document* and *understanding of relevant state policies* are not validated. This confirms the low outcomes for these two factors in the case study meta-analysis.

Component 1 and the validity determined the relative importance. However, this importance can be influenced by the outcomes of the codes (component 2) and the values (component 3). These outcomes provide slight differences in directions of relative importance. As result, some factors were ranked higher or lower than in the estimated importance outcomes. This resulted in the set of preconditions described in the next section.

Looking at the categories, it is difficult to compare them because of the differences between the CSFs and not having an equal number of factors in each category. However, two categories are noteworthy. Category 6 – Support activities – scored highest when looking at Table 14 indicating that this category is important for CBET. Category 5 – External environment – scored lowest indicating that this category maybe not that important.

Box 7. Conclusion Synthesis – Relative importance.

To conclude, the following factors are considered as relative important: *accountability, autonomy, alternative land use resources, infrastructure,* and *funding* followed by the other 12 CSFs (see Table 14) that needs to be taken into account based on their moderately importance and positive influence. These are validated by the survey and interview responses. The factors that seem not that important are *project plans as working document* and *understanding of relevant state policies* because these are not validated by the survey and/or interviews confirming the low score on their estimated importance. Therefore, these two factors were excluded in the set of preconditions. The ranking of the relative importance is eventually determined by the end results of the three components.

Table 14. Synthesis. If a cell is empty, no striking outcome was found. The symbol / indicates that two striking outcomes were found for the particular CSF. Symbols ++ or -- indicate if the factors are considered as an outlier.

| Category | Critical Success Factor | | Case study met | a-analysis + In-d | epth analysis | |
|--------------------------|--|-------------|----------------|-------------------|---------------|-----------|
| | | Importance | Influence | Influence | Validity | Validity |
| | | scale | scale | scale | scale | scale |
| | | Component 1 | Component 2 | Component 3 | Survey | Interview |
| | T.C. | (Frequency) | (Codes) | (values) | (validity) | (vanaity) |
| (1) Resource system | Infrastructure | + | + | +/+- | + | +- |
| characteristics | Attractiveness of the environment | - | | | + | +- |
| | Alternative land use resources | + | +/- | +/+- | +- | + |
| (2) Group | Appropriate leadership | - | - | | + | +- |
| characteristics | Wide awareness and knowledge of the | +- | | | + | + |
| | initiative | | | | + | |
| | Local interests and motivation | +- | | - | + | +- |
| | Cultural and socio-economic differences | +- | +/- | | + | +- |
| | in harmony | | | | | |
| (3) Relationship | Fair allocation of benefits | +- | + | - | + | + |
| between category $1 + 2$ | | | | | | |
| (4) Institutional | Accountability | + | +/- | ++/+- | + | + |
| arrangements | Communication throughout the initiative | +- | | - | + | + |
| | Rule enforcement and appropriate rules- | - | | | + | + |
| | in-one | | | | | |
| | Collaboration and participation in | +- | | | + | +- |
| | decision-making | | | | | |
| (5) External | Understanding of relevant state policies | - | | | +- | - |
| environment | Monitoring and feedback | +- | | | + | +- |
| | Changes in biodiversity | +- | | | + | + |
| | Using project plans as working document | - | | | + | - |
| (6) Support activities | Trainings | +- | | +- | + | + |
| | Funding | + | + | - | + | + |
| | Autonomy | + | +/ | ++ | + | + |

4.3.1. SET OF PRECONDITIONS FOR SUCCESSFUL CBET

Based on the synthesis described above, the following set of preconditions for successful CBET is determined, which are the generated hypotheses for further research (see Table 15).

| Critical Success Factor | | Explanation | | | |
|-------------------------|--------------------------------|---|--|--|--|
| • | Accountability | The enterprise, community, and involved external actors | | | |
| | | are taking responsibility for the management of CBET. | | | |
| • | Autonomy | CBET is self-governed by the enterprise and community | | | |
| | | indicating that they have ownership. | | | |
| • | Infrastructure | The transport, food/drink, and accommodation facilities | | | |
| | | are sufficient for maintaining CBET. | | | |
| • | Alternative land use resources | Other land use resources (e.g. farming) are sustaining | | | |
| | | CBET by providing a second (or first) source of income. | | | |
| • | Funding | The enterprise and community receive funding to facility | | | |
| | | CBET in their development and maintenance. | | | |
| • | Fair allocation of benefits | The benefits obtained by CBET are divided fairly among | | | |
| | | the enterprise and community members. | | | |
| | Trainings | There are good trainings or workshops provided to the | | | |
| | Tunings | enterprise and community for developing their skills to | | | |
| | | develop and maintain CBET. | | | |
| | Cultural and socio-economic | The norms and values of the enterprise and community | | | |
| | differences in harmony | are taken into account resulting in no conflicts, e.g. | | | |
| | differences in narmony | corruption or competing. | | | |
| | Changes in biodiversity | There is a positive impact on biodiversity in the area due | | | |
| | Changes in blourversity | to CBET (e.g. less litter, less deforestation). | | | |
| | Wide ewereness and | The enterprise and community are aware of CBET rules | | | |
| | knowledge of the initiative | and activities and contain the knowledge to develop and | | | |
| | knowledge of the initiative | maintain CBET. | | | |
| | Collaboration and | The enterprise and community are involved in the | | | |
| • | conaboration in decision | decision-making process regarding the development and | | | |
| | participation in decision- | maintenance of CBET. | | | |
| | | The CBET outcomes are monitored and evaluated over a | | | |
| • | Monitoring and feedback | certain period to determine the progress and need for | | | |
| | | improvement. | | | |
| • | Communication throughout | There are regular meetings or consultations to keep the | | | |
| | the initiative | enterprise and community informed. | | | |
| • | Local interests and motivation | The enterprise and community are interested and | | | |
| | . | motivated to participate and be involved in CBET. | | | |
| • | Rule enforcement and | The rules and regulations concerning CBET are easy to | | | |
| | appropriate rules-in-one | understand and followed by a good enforcement system. | | | |
| • | Attractiveness of the | The enterprise and community are located in an area | | | |
| | environment | attractive for tourists increasing their willingness to pay | | | |
| | | for CBET. | | | |
| • | Appropriate leadership | The enterprise and community contain a fair, well- | | | |
| | | connected, and effective leader for CBET. | | | |

Table 15. Set of preconditions for successful community-based ecotourism (CBET).

5. DISCUSSION

This chapter presents the outcome of this research divided among five sections. Firstly, the innovative aspects are highlighted to emphasizes the contribution of this research concerning the theoretical background, methodological approach, and CBET in practice (see section 5.1.). Secondly, the unexpected results are discussed and placed in the context of current literature (see section 5.2.). Thirdly, the limitations of this research are discussed in section 5.3. which led to several future research suggestions presented in section 5.4. Lastly, section 5.5. describes the established policy recommendation for CBET initiatives.

5.1. INNOVATIVE ASPECTS

5.1.1. THEORETICAL IMPLICATIONS

CBET is seen as an alternative bottom-up strategy to offer a livelihood without negatively impacting the environment. However, these initiatives have a high failure rate (Ohl-Schacherer et al., 2008). Therefore, this research explored what preconditions are important for the success of CBET. To investigate this, this research developed a theoretical framework containing 19 CSFs from the fields of entrepreneurship, sustainability, tourism, and governance. These factors were applied to the case study meta-analysis to determine the relative importance of the CSFs by looking at the 'relative weight' and 'direction' of the CSFs within a large-N of case studies. Furthermore, the factors were clustered into six categories to not focus only on the list of factors but to give a direction to possible configurations of the factors that may bear a causal relationship (Agrawal, 2001). However, this needs to be further investigated.

In this research, the CSFs were tested on 50 case studies addressing CBET to gain insights concerning the relative importance, responses from the interviews, and surveys validated these insights. The findings resulted in a set of preconditions for the success of CBET. Therefore, the developed theoretical framework is suitable for exploring and understanding the preconditions for the success of CBET. However, the theoretical framework contains several limitations that are addressed in section 5.3. This was unavoidable since the framework has never been applied before on existing literature or in practice. Therefore, further research is needed to adjust the framework to a more in-depth analysis using different approaches (see section 5.4.). Despite this, this research focused on a global scale to generalize the preconditions which increased the reliability that this framework can be applied on a broad scale of enterprises in a wider context (Byczek, 2011a) and is, therefore, the basis for modifying the theoretical framework further.

5.1.2. METHODOLOGICAL ASPECTS

This research used different quantitative and qualitative methods in the form of triangulation to identify the set of preconditions. By combining the case study meta-analysis with insights gained from a survey and interviews, the reliability and validity of the results have been increased. This is argued in the method section. After conducting this triangulation, it can be concluded that conducting a case study meta-analysis is a suitable contribution to explorative research. In particular, this analysis was suitable for testing a wide network of factors, not explicitly linked to CBET on specific CBET case studies. This presented a robust set of preconditions that can be applied universally in the context of CBET. Therefore, this research provided relevant- and inspiring insights concerning case study meta-analysis in general, which can be used as a basis for identifying correlations and configurations. Furthermore, the use of triangulation made it possible to validate the results with responses obtained from a survey and interviews. The obtained results indicated that the CSFs are responsible for the success of CBET which increased the internal validity (Bryman, 2012). This increased the reliability of the results from the case study meta-analysis. Therefore, this research suggests that triangulation is a suitable method for analyzing complex concepts such as CBET. However, further research is needed to establish this internal validity by identifying configurations and calculating the correlation of the independent variables and the dependent variable.

5.2. UNEXPECTED RESULTS

The CSFs were selected by their impact on the success of community-based enterprises, which were derived from literature claims. These factors provided insights into the success of CBET. Therefore, all factors were expected to be contributing to the success of CBET. However, some unexpected results were found indicating that little, contrasting, or no evidence were found in the case studies about the CSFs and their relevance and relative importance.

The factor *monitoring and feedback* scored lower on relative importance in contrast with the information provided in the literature (i.e. monitoring is often described as an important factor) (Lonn et al., 2018; Mearns, 2011). The factor *fair allocation of benefits* is defined as moderately important while it is a part of the goal of CBET to ensure socio-economic development and environmental management (Kiss, 2004). *Appropriate leadership* is not that important considering the outcomes and contradicting information was provided in the interviews. However, it can still be considered as relative important for specific locations and is therefore ranked as the lowest precondition (Pornprasit & Rurkkhum, 2019).

Alternative land use resources is often discussed in the case studies explaining that CBET is not the main source of income for the local people but that alternative land use resources such as farming and hunting are important to maintaining next to CBET (Nielsen, 2001; Timothy & White, 1999). These alternative resources contribute to CBET, for example, by delivering food or giving cultural workshops and are confirmed by the interviews with JED and WILDTRUST (personal communication, June 13, 2020; personal communication, June 17, 2020). However other case studies mentioned that CBET is the main source of income instead of e.g. farming (Byczek, 2011a; Taylor et al., 2008). Therefore, both interpretations were taken into account during the coding process.

To survey responses, almost all CSFs were validated as important or extremely important. Only *alternative land use resources* and *understanding of relevant state policies* scored neutral. However, the interviews did not validate the factors *using project plans as working document* and *understanding of relevant state policies* as important. Therefore, these two were excluded in the final set of preconditions. Highlighting, in total 17 preconditions contribute to the success of CBET.

Looking at the influence on the relative importance, it was unexpected that four factors showed inconsistency, namely *autonomy, accountability, alternative land use resources*, and *cultural and socio-economic differences in harmony*. This can be explained by the found interlinkages between factors and may influence the success of the management. Thus emphasizing the importance of configuration (Baggio et al., 2016). It was expected for *alternative land use resources* to be inconsistent since the case studies itself discussed contradicting insights, which explains the high score on code (-1). The factor *cultural and socio-economic differences in harmony* remains vague when looking at the contradicted outcomes of code (0) and (-1). This can be explained by the many conflicts addressed in the case studies. Furthermore, more information is required for the factor *changes in biodiversity* because this is the only factor scoring high on conflicting evidence.

The factors *autonomy* and *accountability* were defined as outliers. This is unexpected for *autonomy* when looking at the codes since self-governance is linked to CBET (i.e. the communities control CBET) (Denman, 2001). This factor scored highest on code (-1) indicating that autonomy may influences CBET negatively. However, this can be explained by the found interlinkages. Self-governance is dependent on many other factors such as accountability, funding, leadership, etc. and therefore scored inconsistently. Besides, *Autonomy* scored highest on value (+) indicating that this factor is often discussed and explained in the case studies. In contradiction, the factor *accountability* was expected to be an outlier because of the high scores on frequency, codes, and values and was expected to be inconsistent because of the involvement of external actors in assisting communities (Musavengane & Kloppers, 2020).

To highlight, the external actors were not included in the scope of this research. However, they are indirectly related to CBET and in particular to *accountability, funding, trainings, and autonomy*. This is validated by the interviews. External actors dominate in the factor *accountability* and are therefore taken into account for this factor during the coding process. This indicates that external actors are important concerning the development of CBET. However, it is not explicitly investigated how this involvement of external actors is divided. Future research is necessary for this.

As last, there were some missing factors identified during the coding process, which are considered as moderately important for the success of CBET, namely *marketing, governmental support, gender inequality,* and *market structure*. The factor *marketing* is often mentioned in interviews as well. According to TEA (personal communication, July 2, 2002), marketing is extremely important to consider for improving the success of CBET but the understanding of marketing models is limited (Donohoe & Needham, 2008). Therefore, marketing needs further investigation to be in balance. Otherwise, CBET may lead to mass tourism, which is a different concept of tourism that has negative impacts on the environment and socio-economic development (Ma et al., 2019a; A. Stronza, 2005). The goal of CBET will be not achieved then.

5.3. LIMITATIONS

In this research, several limitations have been encountered. Firstly, the scope of the research is only focusing on the enterprise and involved community. No results were gained for other stakeholders such as NGOs, Tourists, and Governmental authorities. In the in-depth analysis, it became clear that external actors do have a contribution to the success of CBET by, for example, *accountability, funding, rules,* and *trainings*. External actors were often described as indirectly being responsible for parts of the management. Therefore external actors were included in the coding process for the factor *accountability.* Furthermore, this scope does not include the different stages of CBET (e.g. developing or running stage) and does not address long-term sustainability as a time scale. This is necessary for further research because projects are changing during the years (Denman, 2001; Ngece, 2002). However, the scope of this research increased the reliability and validity of the set of preconditions and made it possible to generalize the outcomes by the coding process (Bryman, 2012).

Secondly, the theoretical framework that was developed may not be complete. During the coding process is this validated because four missing factors were found in the case studies: *marketing, gender inequality, governmental support,* and *market structure*. Highlighting the factor *marketing* as a 'blind spot' for this framework because it was explicitly told during the interviews that this factor is important for the success of CBET (TEA, personal communication, July 2, 2020; JED, personal communication, June 13, 2020). Furthermore, the factors were analyzed separately whereby no

configurations were identified. However, there are a few interlinkages recognized as additional findings and the factors were clustered into six categories based on the clustering from Agrawal (2001).

On the opposite, this framework contained a broad overview of factors from four different fields (see literature review, Figure 2) that were indirectly related to CBET. The factors are selected via a literature review and were narrowed down by looking at the frequency. Furthermore, 19 factors is a suitable number for investigating. Including more factors would make it unclear. More importantly, the framework provided new insights because it has never been applied before.

Lastly, the used methods contain several limitations. Concerning the case study meta-analysis, the coding process involved information bias because of missing data. There was only 1 case study in which all CSFs were found. Therefore, the survey and interviews were used to validate the results (Bryman, 2012). Besides, the coding process identified a trend of trade-offs which decreased the subjectivity and increased the reliability (Bryman, 2012). Furthermore, no standardized approach was possible because each case study addressed a different approach, research question, objective, and CBET definition. Furthermore, each case study defined success differently. This substantiates why no correlation was possible between the CSFs and the actual success of CBET. However, the results present a robust set of preconditions based on descriptive statistics, including the boxplots. To attempt standardizing the conceptualization of CBET, seven criteria have been used from Honey (2008) and were revised into more comprehensive criteria.

Concerning the survey and semi-structured interviews, self-reporting led to bias in the survey responses because participants were required to respond to closed questions without interference (Bryman, 2012). The participants were not randomized selected but involved participants that were interested in this research. Furthermore, the responses are a small-N compared to the case studies. The interviews were limited by language barriers and are a small-N either. However, the focus is on case study meta-analysis, the survey and interviews were only used for validation. These small-N's are understandable due to the current situation of the pandemic affecting the tourism industry negatively. As last, the results of the case study meta-analysis and survey are pseudo precision. However, the consistency and transparency of the coding process provided accurate results.

5.4. FUTURE RESEARCH

This research aimed at exploring the CSFs to identify a set of preconditions for successful CBET. The obtained results provided several insights and building blocks for future research. Based on these insights the following theoretical avenues were made on which new research topics can be identified.

- This exploratory research generated hypotheses, which resulted in the set of preconditions. Further research is necessary to test these hypotheses by using, for example, natural/quasiexperiments. This kind of research investigates the causal impact of an intervention. Therefore, these experiments are suitable to measure the actual success of CBET enterprises (Bryman, 2012). After defining the actual success, a correlation can be calculated based on the scores of the CSFs and the actual success by for example an Exploratory Factor Analysis (Bryman, 2012).
- This exploratory research highlighted the need for identifying the importance of configurations. This will increase the insights in the preconditions for successful CBET. The factors that often go together in successful case studies will be determined by clustering the relative importance of the CSFs. The categories and interlinkages addressed in this research indicate already the first step to defining the configurations. Qualitative Comparative Analysis can be used (Baggio et al., 2016).

• The theoretical framework represents the CSFs and can be elaborated by adding the missing factors and by investigating the factors on different time scales (i.e. short-term vs. long-term perspective to sustain CBET), different stages (i.e. when CBET is established, still developing, or in the running), and different geographical scales (i.e. local, regional and national scale). Furthermore, more accurate data can be generated when the factors will be analyzed from the external actor's perspective (e.g. NGOs, and governmental authorities) because more stakeholders are involved in CBET.

5.5. POLICY RECOMMENDATIONS

The results of this research provided policy advice concerning the CSFs, the set of preconditions, and additional findings. This advice is made for existing CBET enterprises and communities and can be useful for other stakeholders to understand the perspective of the involved case studies.

- The set of preconditions for successful CBET was created for recommending the relative CSFs to existing CBET enterprises and communities. Therefore, to contribute this knowledge to the case studies that participated in this research, an infographic was made (see Figure 16). This is an informative illustration to present the set of preconditions in an understandable and simplified way. The advice that follows from this: follow this set of preconditions to see if improvements can be made. This set (may) figures as a guideline to maintain, or sustaining the success of CBET. Note that the missing factor marketing is included in this infographic because it is considered as a 'blind spot' in the literature.
- Following the results of the in-depth analysis, it is recommended to focus on the management of CBET concerning *autonomy*, *accountability*, and *alternative land use resources* because of inconsistency. These inconsistent results may indicate that the biggest challenges are related to these factors. Therefore extra attention is required on these factors by the enterprise and community.
- Marketing is considered as a missing factor to achieve the goal of CBET. However, it is recommended to balance the promotion of CBET in a way that it will not lead to mass tourism because this leads to negative impacts on the biodiversity and local people, the CBET goal will not be achieved then (Ma et al., 2019a; A. Stronza, 2005).
- Following the results of the CBET definition, it is recommended to look at the following criteria for recognizing CBET: (1) involves travel to natural destinations, (2) minimizes impacts, (3) builds environmental awareness, (4) provides benefits for conservation, (5) provides benefits (and empowerment) for the local people and (6) involves the local culture. These criteria were validated as suitable aspects of CBET and may be useful to standardize the conceptualization of CBET. By providing a more comprehensive understanding of CBET, the case studies can discover new insights on where the enterprise should focus on concerning the management to increase its success.



Figure 16. Infographic - set of preconditions for the success of CBET.

6. CONCLUSION

This research aimed to create a set of preconditions for successful community-based ecotourism by analyzing the relative importance of the critical success factors and to make recommendations based on the set of preconditions. Eventually, this exploratory research provided insights on the relative importance which is defined by three components:

- (1) The frequency of the CSF provided an estimation of the 'relative weight' of each factor;
- (2) The frequency of codes provided the 'relative direction' indicating if the importance is influenced positively or negatively by each factor; and
- (3) The frequency of the values indicated the 'relative direction' as well.
- (4) Eventually, the relative importance was ranked by using different scales in a synthesis.

These steps were followed to create the set of preconditions, which answered the research question for this research: *What factors contribute to the success of community-based ecotourism?*

To explore this question, 19 critical success factors (CSFs) were selected and ordered in six categories. These factors were expected to contributing to successful CBET and were applied on 50 case studies regarding CBET explicitly to validate if they are suitable as a precondition. For this, a case study meta-analysis was conducted combined with a survey and interviews to validate the results of the three components. This led to the following main results divided per category.

Category 1 – Resource system characteristics – contained two factors that were estimated as important: *infrastructure* and *alternative land use resources*, one factor as moderately important.

Category 2 – Group characteristics contained no factors estimated as important and three as moderately important.

Category 3 – Relationship between category 1 + 2 – estimated one factor as moderately important.

Category 4 – Institutional arrangements – estimated one factor as important: *accountability* and two other factors as moderately important.

Category 5 – External environment – contained no factors estimated as important and two factors as moderately important.

Category 6 – Support activities – contained two factors that were estimated as important: *autonomy* and *funding* and one factor as moderately important.

This indicated that 5 factors were estimated as important based on the 'relative weighted', followed by 9 factors as moderately important and 5 as not that important. These outcomes were influenced positively or negatively by the 'relative direction' of the codes and values. This slightly changed the ranking of the factors. As last, the relative importance is checked on validity. This emerged into the synthesis presenting the main results (see Table 14). The factors *accountability, autonomy, alternative land use resources, infrastructure,* and *funding* were ranked highest on relative importance. Followed by the other 12 CSFs, considered as relative important based on their moderate importance and positive influence. these were validated by the survey and interview responses. The factors *project*

plans as working document and *understanding of relevant state policies* were not validated as an important factor and therefore, these two factors were excluded in the set of preconditions. The ranking of the relative importance is eventually determined by the results of the three components.

These findings resulted in the set of preconditions, i.e. set of generated hypotheses, and contain 17 critical success factors that were validated in the contribution to successful CBET (see Table 15). Hereby, the research question was answered. The 5 factors that scored highest are *accountability* followed by *autonomy, infrastructure, alternative land use resources*, and *funding*. These 17 factors contribute to the success of CBET. However, some additional findings were found concerning the CBET definition, missing factors, and interlinkages recognizing configurations. These results were obtained to make recommendations. However, the set of preconditions indicates which CSFs are necessary for endeavoring successful CBET and is therefore a reliable first step. Further research is necessary to test the generated hypotheses, to calculate the correlation between the factors and the actual success, and to identify the configuration between the factors providing new insights concerning the success of CBET enterprises.

REFERENCES

- Afenyo, E. A., & Amuquandoh, F. E. (2014). Who Benefits from Community-based Ecotourism Development? Insights from Tafi Atome, Ghana. *Tourism Planning and Development*, *11*(2), 179–190. https://doi.org/10.1080/21568316.2013.864994
- Agrawal, A. (2001). Common property institutions and sustainable governance of resources. *World Development*, *29*(10), 1649–1672. https://doi.org/10.1016/S0305-750X(01)00063-8
- Anup, K. C. (2017). Ecotourism in Nepal. *The Gaze: Journal of Tourism and Hospitality*, *8*, 1–19. https://doi.org/10.3126/gaze.v8i0.17827
- Asadi, A., & Kohan, M. F. Z. (2011). The role of Entrepreneurship on Ecotourism development. 10, 38–42. http://www.ipedr.com/vol10/8-E00015.pdf
- Baggio, J. A., Barnett, A. J., Perez-Ibara, I., Brady, U., Ratajczyk, E., Rollins, N., Rubiños, C., Shin, H. C., Yu, D. J., Aggarwal, R., Anderies, J. M., & Janssen, M. A. (2016). Explaining success and failure in the commons: The configural nature of Ostrom's institutional design principles. *International Journal of the Commons*, 10(2), 417–439. https://doi.org/10.18352/ijc.634
- Baloch, M. A., Danish, Khan, S. U. D., Ulucak, Z. Ş., & Ahmad, A. (2020). Analyzing the relationship between poverty, income inequality, and CO2 emission in Sub-Saharan African countries. *Science of the Total Environment*, 740. https://doi.org/10.1016/j.scitotenv.2020.139867
- Baral, N., & Stern, M. J. (2009). Looking back and looking ahead: Local empowerment and governance in the Annapurna Conservation Area, Nepal. *Environmental Conservation*, 37(1), 54–63. https://doi.org/10.1017/S0376892909990269
- Barnes, C., Claus, R., Driessen, P., Dos Santos, M. J. F., George, M. A., & Van Laerhoven, F. (2017). Uniting forest and livelihood outcomes? Analyzing external actor interventions in sustainable livelihoods in a community forest management context. *International Journal of the Commons*, 11(1), 532–571. https://doi.org/10.18352/ijc.750
- Barnes, C., & van Laerhoven, F. (2015). Making it last? Analysing the role of NGO interventions in the development of institutions for durable collective action in Indian community forestry. *Environmental Science and Policy*, 53, 192–205. https://doi.org/10.1016/j.envsci.2014.06.008
- Belsky, J. M. (1999). Misrepresenting communities: The politics of community-based rural ecotourism in Gales Point Manatee, Belize. *Rural Sociology*, 64(4), 641–666. https://doi.org/10.1111/j.1549-0831.1999.tb00382.x
- Bennett, E. M., Cramer, W., Begossi, A., Cundill, G., Díaz, S., Egoh, B. N., Geijzendorffer, I. R., Krug, C. B., Lavorel, S., Lazos, E., Lebel, L., Martín-López, B., Meyfroidt, P., Mooney, H. A., Nel, J. L., Pascual, U., Payet, K., Harguindeguy, N. P., Peterson, G. D., ... Woodward, G. (2015). Linking biodiversity, ecosystem services, and human well-being: three challenges for designing research for sustainability. *Current Opinion in Environmental Sustainability*, *14*, 76–85. https://doi.org/10.1016/j.cosust.2015.03.007
- Berkes, F., & Davidson-Hunt, I. J. (2010). Innovating through commons use: community-based enterprises. International Journal of the Commons, 4(1), 1–7. https://doi.org/10.18352/ijc.206
- Bhalla, P., Coghlan, A., & Bhattacharya, P. (2016). Homestays' contribution to community-based ecotourism in the Himalayan region of India. *Tourism Recreation Research*, 41(2), 213–228. https://doi.org/10.1080/02508281.2016.1178474
- Bryman, A. (2012). Social Research Methods (4th ed.). Oxford University Press.
- Byczek, C. (2011a). Blessings for All? Community-Based Ecotourism in Bali Between Global, National, and Local Interests A Case Study. *Aseas*, 4(1), 81–106. https://doi.org/10.1074/jbc.M110.101808

- Byczek, C. (2011b). Blessings for All? Community-Based Ecotourism in Bali Between Global, National, and Local Interests A Case Study. *Aseas*, 4(1), 81–106.
- Catalano, A. S., Lyons-White, J., Mills, M. M., & Knight, A. T. (2019). Learning from published project failures in conservation. *Biological Conservation*, 238(April), 108223. https://doi.org/10.1016/j.biocon.2019.108223
- Clarke, A. (1999). A practical use of key success factors to improve the effectiveness of project management. International Journal of Project Management, 17(3), 139–145. https://doi.org/10.1016/S0263-7863(98)00031-3
- Cox, M., Arnold, G., & Tomás, S. V. (2010). A review of design principles for community-based natural resource management. *Ecology and Society*, *15*(4). https://doi.org/10.5751/ES-03704-150438
- De Koning, M., Nguyen, T., Lockwood, M., Sengchanthavong, S., & Phommasane, S. (2017). Collaborative Governance of Protected Areas: Success Factors and Prospects for Hin Nam No National Protected Area, Central Laos. *Conservation and Society*, *15*(1), 87–99. https://doi.org/10.4103/0972-4923.201396
- Denman, R. (2001). Guidelines for community-based ecotourism development. In WWF international (Issue July). https://doi.org/10.1002/1522-1970(200101/02)3:1<72::AID-JTR276>3.0.CO;2-4
- Dodds, R., Ali, A., & Galaski, K. (2018). Mobilizing knowledge: determining key elements for success and pitfalls in developing community-based tourism. *Current Issues in Tourism*, *21*(13), 1547–1568. https://doi.org/10.1080/13683500.2016.1150257
- Donohoe, H. M., & Needham, R. D. (2008). Internet-based ecotourism marketing: Evaluating Canadian sensitivity to ecotourism tenets. *Journal of Ecotourism*, 7(1), 15–43. https://doi.org/10.2167/joe185.0
- Đukić, V., Volić, I., Tišma, S., & Jelinčić, D. A. (2014). Responsible Community Based Ecotourism Initiatives in Protected Rural Areas of the Balkans: Case Studies from Serbia and Croatia. American Journal of Tourism Management, 3(1B), 51–63. https://doi.org/10.5923/s.tourism.201402.06
- Eshun, G., Naana, J., & Tonto, P. (2014). Community-based ecotourism: Its socio-economic impactsat Boabeng-Fiema Monkey Sanctuary, Ghana. 26(26), 67–81.
- Farrelly, T. A. (2011). Indigenous and democratic decision-making: Issues from community-based ecotourism in the boumā National Heritage Park, Fiji. *Journal of Sustainable Tourism*, 19(7), 817–835. https://doi.org/10.1080/09669582.2011.553390
- Fiorello, A., & Bo, D. (2012). Community-Based Ecotourism to Meet the New Tourist's Expectations: An Exploratory Study. *Journal of Hospitality Marketing and Management*, 21(7), 758–778. https://doi.org/10.1080/19368623.2012.624293
- Foucat, V. S. A. (2002). Community-based ecotourism management moving towards sustainability, in Ventanilla, Oaxaca, Mexico. Ocean and Coastal Management, 45(8), 511–529. https://doi.org/10.1016/S0964-5691(02)00083-2
- Ghasemi, M., & Hamzah, A. (2013). ASSESSING THE EXECUTIVE SUCCESS OF NON-GOVERNMENTAL ORGANIZATIONS IN IMPLEMENTING COMMUNITY-BASED ECOTOURISM PROJECTS: A CASE STUDY OF MALAYSIAN NATURE SOCIETY (MNS) WITHIN ORANG ASLI COMMUNITY AT ULU GEROH VILLAGE, GOPENG, MALAYSIA. *BIMP-EAGA Journal for Sustainable Tourism Development*, 2(2), 2011–2014.
- Graham, S., Metcalf, A. L., Gill, N., Niemiec, R., Moreno, C., Bach, T., Ikutegbe, V., Hallstrom, L., Ma, Z., & Lubeck, A. (2019). Opportunities for better use of collective action theory in research and governance for invasive species management. *Conservation Biology*, 33(2), 275–287. https://doi.org/10.1111/cobi.13266
- Gray, B., Kirkwood, J., Etemaddar, M., & Monahan, E. (2018). Sustainable business models for communitybased enterprises in Samoa and Tonga. *Small Enterprise Research*, *25*(2), 99–113. https://doi.org/10.1080/13215906.2018.1479293
- Grigg, J. (2016). Is ecotourism an effective conservation tool? PLOS Early Career Research Community.

https://blogs.plos.org/thestudentblog/2016/05/20/is-ecotourism-an-effective-conservation-tool/

- Gruber, J., Mbatu, R., Johns, R., & Dixon, B. (2018). Measuring conservation success beyond the traditional biological criteria: the case of conservation projects in Costa Rica, Mekong Valley, and Cameroon. *Natural Resources Forum*, 42(1), 19–31. https://doi.org/10.1111/1477-8947.12132
- Gumede, T. K., & Nzama, A. T. (2019). Ecotourism as a mechanism for local economic development: The case of communities adjacent to the Oribi Gorge Nature Reserve, KwaZulu-Natal, South Africa. *African Journal of Hospitality, Tourism and Leisure*, 8(4), 1–19.
- Gurău, C., & Dana, L. P. (2018). Environmentally-driven community entrepreneurship: Mapping the link between natural environment, local community and entrepreneurship. *Technological Forecasting and Social Change*, *129*(November 2017), 221–231. https://doi.org/10.1016/j.techfore.2017.11.023
- Haenn, N., Harnish, A., & Wilk, R. (2016). *The Environment in Anthropology: A Reader in Ecology, culture and sustainable living*. New York University Press.
- Hamzah, A., & Mohamad, N. H. (2012). Critical success factors of community based ecotourism: Case study of miso walaihomestay, Kinabatangan, Sabah. *The Malaysian Forester*, *75*(1), 27–42.
- Hasan, M. B., Driessen, P., Zoomers, A., & Van Laerhoven, F. (2020). How can NGOs support collective action among the users of rural drinking water systems? A case study of Managed Aquifer Recharge (MAR) systems in Bangladesh. *World Development*, *126*, 104710. https://doi.org/10.1016/j.worlddev.2019.104710
- Heil, E. J. (2017). Through the eyes of Asa; the role of integrated conservation development programs in Ghandruk, Nepal. World Journal of Environmental Research, 7(1), 17–26. https://doi.org/10.18844/wjer.v7i1.2386
- Holden, A. (2010). Exploring stakeholders' perceptions of sustainable tourism development in the annapurna conservation area: Issues and challenge. *Tourism and Hospitality, Planning and Development*, 7(4), 337– 351. https://doi.org/10.1080/1479053X.2010.523177
- Holguín, C., Alejandro, D., & Mara, R. B. (2014). Networks in Strengthening Community-based Ecotourism in the Sierra Norte of Oaxaca, Mexico International Journal Advances in Social Science and Humanities Available online at : www.ijassh.com Networks in Strengthening Community-based Ecotourism in the Si. May.
- Jamaliah, M. M., & Powell, R. B. (2018). Ecotourism resilience to climate change in Dana Biosphere Reserve, Jordan. Journal of Sustainable Tourism, 26(4), 519–536. https://doi.org/10.1080/09669582.2017.1360893
- Jones, S. (2005). Community-based ecotourism: The significance of social capital. Annals of Tourism Research, 32(2), 303–324. https://doi.org/10.1016/j.annals.2004.06.007
- Keeble, B. R. (1988). The Brundtland Report: "Our Common Future." In *Medicine and War* (Vol. 4, Issue 1). https://doi.org/10.1080/07488008808408783
- Khan, K. S., Kunz, R., Kleijnen, J., & Antes, G. (2003). Five steps to conducting a systemstic review. *Journal of the Royal Society of Medicine*, *96*, 118–121.
- Kim, K. H., & Park, D. B. (2017). Relationships Among Perceived Value, Satisfaction, and Loyalty: Community-Based Ecotourism in Korea. *Journal of Travel and Tourism Marketing*, 34(2), 171–191. https://doi.org/10.1080/10548408.2016.1156609
- Kiss, A. (2004). Is community-based ecotourism a good use of biodiversity conservation funds? *Trends in Ecology and Evolution*, *19*(5), 232–237. https://doi.org/10.1016/j.tree.2004.03.010
- Koens, J. F., Dieperink, C., & Miranda, M. (2009). Ecotourism as a development strategy: Experiences from Costa Rica. *Environment, Development and Sustainability*, 11(6), 1225–1237. https://doi.org/10.1007/s10668-009-9214-3

- Kontogeorgopoulos, N. (2005). Community-based ecotourism in phuket and ao phangnga, Thailand: Partial victories and bittersweet remedies. *Journal of Sustainable Tourism*, *13*(1), 4–23. https://doi.org/10.1080/17501220508668470
- Lee, D. E., & Bond, M. L. (2018). Quantifying the ecological success of a community-based wildlife conservation area in Tanzania. *Journal of Mammalogy*, *99*(2), 459–464. https://doi.org/10.1093/jmammal/gyy014
- Lee, D. R., & Neves, B. (2009). Rural Poverty and Natural Resources: Improving Access and Sustainable Management. *Agricultural Development Economics Division, FAO*.
- Lettinga, N., Jacquet, P. O., André, J. B., Baumand, N., & Chevallier, C. (2020). Environmental adversity is associated with lower investment in collective actions. *PLoS ONE*, *15*(7 July), 1–23. https://doi.org/10.1371/journal.pone.0236715
- Lonn, P., Mizoue, N., Ota, T., Kajisa, T., & Yoshida, S. (2018). Evaluating the Contribution of Community-based Ecotourism (CBET) to Household Income and Livelihood Changes: A Case Study of the Chambok CBET Program in Cambodia. *Ecological Economics*, 151(April), 62–69. https://doi.org/10.1016/j.ecolecon.2018.04.036
- Ma, B., Yin, R., Zheng, J., Wen, Y., & Hou, Y. (2019a). Estimating the social and ecological impact of communitybased ecotourism in giant panda habitats. *Journal of Environmental Management*, 250(February), 109506. https://doi.org/10.1016/j.jenvman.2019.109506
- Ma, B., Yin, R., Zheng, J., Wen, Y., & Hou, Y. (2019b). Estimating the social and ecological impact of communitybased ecotourism in giant panda habitats. *Journal of Environmental Management*, 250(February), 109506. https://doi.org/10.1016/j.jenvman.2019.109506
- Manyara, G., & Jones, E. (2007). Community-based tourism enterprises development in Kenya: An exploration of their potential as avenues of poverty reduction. *Journal of Sustainable Tourism*, *15*(6), 628–644. https://doi.org/10.2167/jost723.0
- McIntosh, S., & Renard, Y. (2010). Placing the commons at the heart of community development: Three case studies of community enterprise in Caribbean islands. *International Journal of the Commons*, 4(1), 160–182. https://doi.org/10.18352/ijc.135
- Mearns, K. F. (2011). Using Sustainable Tourism Indicators to Measure the Sustainability of a Community-Based Ecotourism Venture: Malealea Lodge & Pony Trek Centre, Lesotho. *Tourism Review International*, 15(1), 135–147. https://doi.org/10.3727/154427211x13139345020499
- Measham, T. G., & Lumbasi, J. A. (2013). Success factors for community-based natural resource management (CBNRM): Lessons from Kenya and Australia. *Environmental Management*, *52*(3), 649–659. https://doi.org/10.1007/s00267-013-0114-9
- Mengist, W., Soromessa, T., & Legese, G. (2020). Method for conducting systematic literature review and metaanalysis for environmental science research. *MethodsX*, 7, 100777. https://doi.org/10.1016/j.mex.2019.100777
- Mensah, I. (2017). Benefits and challenges of community-based ecotourism in park-fringe communities: The case of mesomagor of Kakum National Park, Ghana. *Tourism Review International*, *21*(1), 81–98. https://doi.org/10.3727/154427217X14866652018947
- Mohd Shahwahid, H. O., Mohd Iqbal, M. N., Amira Mas Ayu, A. M., & Farah, M. S. (2013). Assessing service quality of community-based ecotourism: A case study from Kampung Kuantan Firefly Park. *Journal of Tropical Forest Science*, *25*(1), 22–33.

Morgenstern, J. (2018). Survival bias. First10EM Blog. https://first10em.com/ebm/survival-bias/

Mtapuri, O., & Giampiccoli, A. (2019). Tourism, community-based tourism and ecotourism: a definitional problematic. *South African Geographical Journal*, *101*(1), 22–35. https://doi.org/10.1080/03736245.2018.1522598

- Musavengane, R. (2019). Land reform and the promotion of collaborative community-based ecotourism at Somkhanda Game Reserve, South Africa. *Positive Tourism in Africa, March 2019*, 23–35. https://doi.org/10.4324/9780429428685-3
- Musavengane, R., & Kloppers, R. (2020). Social capital: An investment towards community resilience in the collaborative natural resources management of community-based tourism schemes. *Tourism Management Perspectives*, *34*(March 2019), 100654. https://doi.org/10.1016/j.tmp.2020.100654
- Musavengane, R., & Simatele, D. (2017). Significance of social capital in collaborative management of natural resources in Sub-Saharan African rural communities: a qualitative meta-analysis. *South African Geographical Journal*, *99*(3), 267–282. https://doi.org/10.1080/03736245.2016.1231628
- National Geographic. (2019). Conservation. Conservation is the act of protecting Earth's natural resources for current and future generations. https://www.nationalgeographic.org/encyclopedia/conservation/
- Nelson, F. (2004). The evolution and impacts of community-based ecotourism in northern Tanzania. International Institute for Environment and Development, 131, 1–46.
- Ngece, K. (2002). Community Based Ecotourism: What can the people of East Africa learn from success stories elsewhere? *East African Ecotourism Development and Conservation Consultants, November*, 1–4.
- Nielsen, E. a. (2001). Community-Based Ecotourism Development and Management in the Rio Platano Man and the Biosphere Reserve, Honduras Submitted to: The Nature Conservancy International Conservation Program Submitted by:
- Nyaupane, G. P., & Thapa, B. (2004). Evaluation of ecotourism: A comparative assessment in the Annapurna Conservation Area Project, Nepal. *Journal of Ecotourism*, *3*(1), 20–45. https://doi.org/10.1080/14724040408668148
- Ofori, D. F. (2013). Project Management Practices and Critical Success Factors–A Developing Country Perspective. *International Journal of Business and Management*, 8(21), 14–31. https://doi.org/10.5539/ijbm.v8n21p14
- Ohl-Schacherer, J., Mannigel, E., Kirkby, C., Shepard, G. H., & Yu, D. W. (2008). Indigenous ecotourism in the Amazon: A case study of Casa Matsiguenka in Manu National Park, Peru. *Environmental Conservation*, *35*(1), 14–25. https://doi.org/10.1017/S0376892908004517
- Ortega-Álvarez, R., & Calderón-Parra, R. (2020). Linking biological monitoring and wildlife ecotourism: a call for development of comprehensive community-based projects in search of sustainability. *Environment, Development and Sustainability, 0123456789.* https://doi.org/10.1007/s10668-020-00761-7
- Parker, S., & Khare, A. (2005). Understanding success factors for ensuring sustainability in ecotourism development in southern Africa. *Journal of Ecotourism*, *4*(1), 32–46. https://doi.org/10.1080/14724040508668436
- Peredo, A. M., Chrisman, J. J., & Chrisman, T. T. (2006). Toward a Theory of Community-Based Enterprise TOWARD A THEORY OF COMMUNITY-BASED ENTERPRISE ANA MAR?A PEREDO. *Source: The Academy of Management Review Academy of Management Review*, *31*(2), 309–328. http://www.jstor.org/stable/20159204%0Ahttp://about.jstor.org/terms
- Pornprasit, P., & Rurkkhum, S. (2019). Performance evaluation of community-based ecotourism: a case study in Satun province, Thailand. *Journal of Ecotourism*, *18*(1), 42–59. https://doi.org/10.1080/14724049.2017.1379529
- Reimer, J. K. K., & Walter, P. (2013). How do you know it when you see it? Community-based ecotourism in the Cardamom Mountains of southwestern Cambodia. *Tourism Management*, *34*, 122–132. https://doi.org/10.1016/j.tourman.2012.04.002
- Rhama, B. (2020). The meta-analysis of ecotourism in National Parks. *African Journal of Hospitality, Tourism and Leisure*, 9(1), 1–17.

- Romero-Brito, T. P., Buckley, R. C., & Byrne, J. (2016). NGO partnerships in using ecotourism for conservation: Systematic review and meta-analysis. *PLoS ONE*, *11*(11), 1–19. https://doi.org/10.1371/journal.pone.0166919
- Runhaar, H., & Polman, N. (2018). Partnering for nature conservation: NGO-farmer collaboration for meadow bird protection in the Netherlands. *Land Use Policy*, *73*(August 2017), 11–19. https://doi.org/10.1016/j.landusepol.2018.01.033
- Sakata, H., & Prideaux, B. (2013). An alternative approach to community-based ecotourism: A bottom-up locally initiated non-monetised project in Papua New Guinea. *Journal of Sustainable Tourism*, 21(6), 880– 899. https://doi.org/10.1080/09669582.2012.756493
- Smit, E. (n.d.). Bottom -up or the bottom line? Exploring community participation as a viable business strategy for ecotourism operators in the Daintree Coast, Australia. *Business Strategy and the Environmen*. http://mc.manuscriptcentral.com/bse%0D
- Snyman, S. (2017). The role of private sector ecotourism in local socio-economic development in southern Africa. *Journal of Ecotourism, 16*(3), 247–268. https://doi.org/10.1080/14724049.2016.1226318
- Stanley, N. B. (2014). Three-Tiered Approach to Ecotourism Research: Meta-Analysis, Statisical Analysis, and Field Study (Master's thesis). [Texas State University]. https://digital.library.txstate.edu/handle/10877/5270
- Steffen, W., Richardson, K., Rockström, J., Cornell, S. E., Fetzer, I., Bennett, E. M., Biggs, R., Carpenter, S. R., De Vries, W., De Wit, C. A., Folke, C., Gerten, D., Heinke, J., Mace, G. M., Persson, L. M., Ramanathan, V., Reyers, B., & Sörlin, S. (2015). Planetary boundaries: Guiding human development on a changing planet. *Science*, 347(6223). https://doi.org/10.1126/science.1259855
- Sterling, E. J., Betley, E., Sigouin, A., Gomez, A., Toomey, A., Cullman, G., Malone, C., Pekor, A., Arengo, F., Blair, M., Filardi, C., Landrigan, K., & Porzecanski, A. L. (2017). Assessing the evidence for stakeholder engagement in biodiversity conservation. *Biological Conservation*, 209, 159–171. https://doi.org/10.1016/j.biocon.2017.02.008
- Stone, M. T. (2015). Community-based ecotourism: A collaborative partnerships perspective. *Journal of Ecotourism*, 14(2–3), 166–184. https://doi.org/10.1080/14724049.2015.1023309
- Stronza, A. (2005). HOSTS AND HOSTS: THE ANTHROPOLOGY OF COMMUNITY-BASED ECOTOURISM IN THE PERUVIAN AMAZON. *NAPA Bulletin, 23,* 170–190.
- Stronza, A. L., Hunt, C. A., & Fitzgerald, L. A. (2019). Ecotourism for Conservation? *Annual Review of Environment and Resources*, 44(1), 229–253. https://doi.org/10.1146/annurev-environ-101718-033046
- Stronza, Amanda Lee. (2010). Commons management and ecotourism: Ethnographic evidence from the Amazon. *International Journal of the Commons*, 4(1), 56–77. https://doi.org/10.18352/ijc.137
- Taylor, P., Cusack, D., & Dixon, L. (2008). Comminity-Based Ecotourism and Sustainability Community-Based Ecotourism and Sustainability : Cases in Bocas del Toro Province , Panama and Talamanca , Costa Rica. *Journal of Sustainable Forestry*, 22(1–2), 157–182. https://doi.org/10.1300/J091v22n01
- Thakadu, O. T. (2005). Success factors in community based natural resources management in northern Botswana: Lessons from practice. *Natural Resources Forum*, *29*(3), 199–212. https://doi.org/10.1111/j.1477-8947.2005.00130.x
- Timothy, D. J., & White, K. (1999). Community-based ecotourism development on the periphery of Belize. *Current Issues in Tourism*, 2(2–3), 226–242. https://doi.org/10.1080/13683509908667853
- Treephan, P., Visuthismajarn, P., & Isaramalai, S. A. (2019). A model of participatory community-based ecotourism and mangrove forest conservation in Ban Hua Thang, Thailand. *African Journal of Hospitality, Tourism and Leisure*, *8*(5), 1–8.

- Ullah, Z., Badshah, S. I., & Shaheen, M. (2017). STRATEGIC ANALYSIS OF SELECTED CASE STUDIES FOR THEIR. Journal of Business and Tourism, 221–233.
- United Nations. (2019). UN Report: Nature's Dangerous Decline 'Unprecedented'; Species Extinction Rates 'Accelerating.' https://www.un.org/sustainabledevelopment/blog/2019/05/nature-declineunprecedented-report/
- United Nations. (2020). About the Sustainable Development Goals. https://www.un.org/sustainabledevelopment/sustainable-development-goals/
- Ven, S. (2016). Residents' Participation, Perceived Impacts, and Support for Community-based Ecotourism in Cambodia: A Latent Profile Analysis. Asia Pacific Journal of Tourism Research, 21(8), 836–861. https://doi.org/10.1080/10941665.2015.1075565
- Verschuren, P., & Doorewaard, H. (2010). *Designing a Research Project* (Second). Eleven International Publishing.
- Walter, P. (2009). Local knowledge and adult learning in environmental adult education: Community-based ecotourism in southern Thailand. *International Journal of Lifelong Education*, *28*(4), 513–532. https://doi.org/10.1080/02601370903031363
- Walter, P., Regmi, K. D., & Khanal, P. R. (2018). Host learning in community-based ecotourism in Nepal: The case of Sirubari and Ghalegaun homestays. *Tourism Management Perspectives*, *26*(February), 49–58. https://doi.org/10.1016/j.tmp.2018.02.002
- Wondirad, A. (2019). Does ecotourism contribute to sustainable destination development, or is it just a marketing hoax? Analyzing twenty-five years contested journey of ecotourism through a meta-analysis of tourism journal publications. Asia Pacific Journal of Tourism Research, 24(11), 1047–1065. https://doi.org/10.1080/10941665.2019.1665557
- WONDIRAD, A., Tolkach, D., & King, B. (2020). Stakeholder collaboration as a major factor for sustainable ecotourism development in developing countries. *Tourism Management*, *78*(November 2018), 104024. https://doi.org/10.1016/j.tourman.2019.104024
- World Wide Fund For Nature. (n.d.). *WWF has more than 3000 projects on-the-go at any one time*. Retrieved October 5, 2020, from https://wwf.panda.org/knowledge_hub/where_we_work/project/
- Yalegama, S., Chileshe, N., & Ma, T. (2016). Critical success factors for community-driven development projects: A Sri Lankan community perspective. *International Journal of Project Management*, *34*(4), 643–659. https://doi.org/10.1016/j.ijproman.2016.02.006
- Yin, R. K. (2003). Designing Case Studies. In *Qualitative Research Methods* (pp. 359–400). SAGE Publications Ltd. https://doi.org/10.1057/9781137016669

APPENDIX I

| Authors | Explanatory factors |
|--|--|
| Barnes & van | Well-defined boundaries of the resource |
| Laerhoven (2015) | Past successful experiences – social capital |
| · · · · | Meetings |
| | CPR appropriation rules-in-one: |
| | Rules are simple and easy to understand |
| | Locally devised access and management rules |
| | Ease in monitoring and enforcement of rules |
| | Graduated sanctions |
| | Availability of low cost adjudication |
| | Accountability of monitors and other officials to users |
| | Restrictions on harvests matched to regeneration of resources |
| | Understanding of relevant state policies |
| | Wide awareness of CPR institutions and organization |
| | Inclusion of all CPR users' identities and interests |
| | Perceived management canacity of CPR users |
| | Confidence in own canacities |
| | Appropriate leadership |
| | Percention that local authority is not undermined by external actors |
| | Appropriate connections |
| | Sufficient financial and material resources |
| | Confidence that future benefits will be fairly allocated |
| | Supportive external environment |
| Catalana at al. (2010) | Supportive external environment |
| $\frac{\text{Catalano et al. (2019)}}{\text{Clarks (1000)}}$ | Communication throughout the nucleast |
| Clarke (1999) | Clear chiestives and essent |
| | Desching the project into `hite sized shunks' |
| | Breaking the project into bite sized chunks |
| | Using project plans as working documents |
| De Koning et al. | Incentives for community engagement |
| (2017) | Formal mechanisms for power sharing |
| | Local ownership of resources |
| | Downward accountability |
| | Mechanisms for building trust |
| D 11 (0010) | Adaptive approach to performance assessment and improvement |
| Dodds et al. (2018) | Participatory planning and capacity building |
| | Collaboration and partnerships facilitating links to market |
| | Local management/empowerment of community members |
| | Establishment of environmental/community goals |
| | Focus on generating supplemental income for long-term community |
| | sustainability |
| Gruber et al. (2018) | Biological factors: |
| | Changes in the amount of targeted conserved species |
| | Biodiversity |
| | Total area conserved |
| | Environmental factors: |
| | Environmental health |
| | Soil quality |
| | Water quality |
| | Species index |
| | Habitat fragmentation and edge effect |
| | Habitat gap analysis |

Table 16. Summary of explanatory factors found in literature reviews.

| | Institutional/policy factors: |
|--|--|
| | Government policy |
| | Local government organizations |
| | NGOs |
| | Management plans |
| | Funding |
| | Social/cultural/economic factors: |
| | Population change |
| | Urban pressures |
| | Major development projects |
| | Encroachment |
| | Agriculture |
| | Local participation |
| | Culture edge effect |
| | Pastoration |
| | Tourism |
| | Conflict factors |
| | Conflict factors: |
| | International magnetiat |
| | International conflict |
| | Kerugee state |
| | Local/regional conflict |
| | Conflict over resources |
| Hamzah & Mohamad | Dynamic leadership and organization |
| (2012) | The setting up of a tourism cooperative |
| | Partnerships with government agencies |
| | Tourism industry players |
| | Strong commitment to biodiversity conservation |
| Hasan et al. (2020) | Requirements for day-to-day collective action: |
| | Regular meetings |
| | Presence of rules: |
| | On entry |
| | On appropriation |
| | Rule enforcement: |
| | Monitor system |
| | Graduated sanctioning system |
| | The monitoring of monitors |
| | Low-cost adjudication |
| | Requirements for long-term collective action: |
| | Understanding of relevant policies |
| | Participation of users in decision-making |
| | Management capacity of resource users |
| | Fair allocation of benefits |
| | Ability of users to pay |
| | |
| | Willingness of users to pay |
| | Willingness of users to pay Awareness of users |
| | Willingness of users to pay Awareness of users Dynamic leadership |
| | Willingness of users to pay Awareness of users Dynamic leadership Supportive external environment |
| Lee & Bond (2018) | Willingness of users to pay Awareness of users Dynamic leadership Supportive external environment Long-term monitoring |
| Lee & Bond (2018) | Willingness of users to pay Awareness of users Dynamic leadership Supportive external environment Long-term monitoring Locally based monitoring |
| Lee & Bond (2018) | Willingness of users to pay Awareness of users Dynamic leadership Supportive external environment Long-term monitoring Locally based monitoring Community sensitization |
| Lee & Bond (2018) Manyara & Jones (2007) | Willingness of users to pay Awareness of users Dynamic leadership Supportive external environment Long-term monitoring Locally based monitoring Community sensitization Awareness |
| Lee & Bond (2018) Manyara & Jones (2007) | Willingness of users to pay Awareness of users Dynamic leadership Supportive external environment Long-term monitoring Locally based monitoring Community sensitization Awareness Community empowerment |
| Lee & Bond (2018) Manyara & Jones (2007) | Willingness of users to pay Awareness of users Dynamic leadership Supportive external environment Long-term monitoring Locally based monitoring Community sensitization Awareness Community empowerment Effective leadership |
| Lee & Bond (2018) Manyara & Jones (2007) | Willingness of users to pay Awareness of users Dynamic leadership Supportive external environment Long-term monitoring Locally based monitoring Community sensitization Awareness Community empowerment Effective leadership Community capacity building |

| Measham & Lumbasi | Failing due to: |
|-----------------------|--|
| (2013) | Top down project initiation |
| | Lack of economic incentives |
| | Lack of autonomy |
| | Incompatible livelihoods |
| | Opportunity costs |
| Ofori (2013) | Communication factors: |
| | Clarity of mission and goals |
| | Effective communication |
| | Effective consultation with project stakeholders |
| | Well-laid out specifications |
| | Realistic cost and time estimates for the project |
| | Commitment factors |
| | Top management support and commitment |
| | Adequate resources for the project |
| | Commitment to standards and regulations to ensure quality |
| | Commitment to client/beneficiary satisfaction |
| | Competency factors: |
| | Competency and experience of the project personnel |
| | Use of superior and appropriate technology for the project |
| | Coordination factors: |
| | Cool leadership |
| | Teamwork |
| | Monitoring and Foodback |
| | Client involvement |
| Darkar & Khara (2005) | Environmental factore: |
| Parker & Khare (2003) | Environmental factors: |
| | Site houndering |
| | Site boundaries Weter |
| | water |
| | Opportunity costs |
| | Community factors: |
| | Community partnersnip |
| | Community definition |
| | Community dialogue |
| | Poverty and social inclusion |
| | Economic factors: |
| | National political environment |
| | Adequate legal systems and security |
| | Infrastructure |
| <u> </u> | Government policy |
| Runhaar & Polman | Alignment of interests and complementarity of the partners |
| (2018) | Motivation derived from meeting peers |
| | Clarity and congruency of objectives |
| | Capabilities and accountability |
| | Communication |
| | Learning and commitment |
| | Reduced transaction costs by partnership |
| | Willingness of the actors to be involved to participate |
| Smit (n.d.) | Ownership and empowerment |
| | Increase support of tourism activities |
| | Equal decision-making and harmonious relationships |
| | Reduce consumptive land-use and change the local's attitudes |
| | Local interests |
| | Enhance the capacities of local stakeholders |
| | Protect the resource at stake |
| | Decisions with more success and less conflict |
|--------------------------|--|
| | Operate in a sustainable way |
| | Better understanding of the stakeholder's needs and expectations |
| | Clarify the goal towards local issues |
| | Strengthen community development |
| | Establish good relations |
| | Empower communities to decide over their own needs |
| | Increase responsibility |
| | Benefits and support for the communities |
| Sterling et al. (2017) | Stakeholder engagement: |
| | Identifying stakeholders |
| | Timing and degree of stakeholder engagement |
| | Recognizing and respecting stakeholder values and institutions |
| | Stakeholder motivation for engagement |
| | Effective leadership |
| | Effective partnerships |
| Thakadu (2005) | Broadly based participation |
| | Credibility and mutual trust |
| | Willingness and readiness |
| | Perceived benefits and their distribution |
| | Socio-economic and cultural stratification |
| Treephan et al. (2019) | Strong relationship between different groups and individuals |
| | Roles and responsibilities are delegated according to the specific skill |
| | set of each community member |
| | Ecotourism management suits the local community lifestyle and is a |
| | reflection of traditional community practices |
| | The conscience of people in the community regarding culture and |
| | lifestyle is used to inform the management process and increase the |
| | level of income generated for the local area |
| | The fertility of natural resources and the surrounding environment as |
| | an identifying factor of the community |
| WONDIRAD et al | Eactors affecting ecotourism stakeholder collaboration |
| (2020) | Poor tourism governance |
| (2020) | Lack of awareness amongst stakeholders about the relevance of |
| | collaboration |
| | Poor culture (tradition) of collaboration in the society |
| | Resource constraints |
| | Lack of trust and mutual understanding amongst ecotourism |
| | stakeholders |
| | Lack of sufficient and sustained discussion and communication |
| | amongst ecotourism stakeholders |
| | The limited size of the ecotourism sector in the country and in the |
| | region receives little attention within local communities and private |
| | ecotourism enterprises due to its smallness in scale |
| | Existence of diverse interests and unhealthy competition amongst |
| | ecotourism stakeholders |
| | Power friction within governmental organizations and amongst |
| | government local communities and private ecotourism enterprises |
| | Conflicts amongst ethnic tribes |
| Yalegama et al. (2016) | Enabling community environment |
| 1 alogunia et al. (2010) | Measurable project management outcomes by village organization |
| | Community project management engagement |
| | community project munugement engagement |

APPENDIX II

| No. | Name case study | Reference | | | | |
|----------------------|--|--|--|--|--|--|
| 1 | Tafia Atome community | (Afenyo & Amuquandoh, 2014) | | | | |
| 2 | Expediciones Sierra Norte | (Holguín et al., 2014) | | | | |
| 3 | REDSJO network | (Holguín et al., 2014) | | | | |
| 4 | Ghalekharka-Sikles ecotrek | (Nyaupane & Thapa, 2004) | | | | |
| 5 | Gales Point Manatee | (Belsky, 1999) | | | | |
| 6 | Binsar Wildlife Sanctuary | (Bhalla et al., 2016) | | | | |
| 7 | Yeak Laom community | (Ven. 2016) | | | | |
| 8 | La Ventanilla | (Foucat, 2002) | | | | |
| 9 | Jaringan Ekowisata Desa (JED) village network | (Byczek, 2011b) | | | | |
| 10 | Bribri project | (Taylor et al., 2008) | | | | |
| 11 | Ngöbe-Buglé community project | (Taylor et al., 2008) | | | | |
| 12 | APROMOVEN | (Taylor et al. (2008)) | | | | |
| 13 | MONSELVA | (Taylor et al., 2008) | | | | |
| 14 | FUNDICCEP | (Taylor et al., 2000) | | | | |
| 15 | The Teribe project | (Taylor et al., 2000) | | | | |
| 16 | Omoliica village | (Further et al., 2000) | | | | |
| 17 | Konačevo village | (Dukic et al., 2014) | | | | |
| 18 | Ropherov Vinage Boaheng-Fiema Monkey Sanctuary | (Eshun et al. 2014) | | | | |
| 19 | Lavena Backpacker Lodge and Coastal Walk | (Earrelly 2011) | | | | |
| 20 | Malaysian Nature Society | (Ghasemi & Hamzah 2013) | | | | |
| 20 | Miso Walai homestay | (Hamzah & Mohamad 2012) | | | | |
| $\frac{21}{22}$ | Casa Matsiguenka lodge | (Ohl-Schacherer et al. 2008) | | | | |
| 22 | Tumani Tenda ecocamp | (Jones 2005) | | | | |
| $\frac{23}{24}$ | Arumchi village | (Kim & Park 2017) | | | | |
| 2 4 25 | Uno Kasiyeonkot village | (Kim & Park, 2017) $(Kim & Park, 2017)$ | | | | |
| 25 | Jumae village | (Kim & Park, 2017) $(Kim & Park, 2017)$ | | | | |
| 20 | Vupo village | (Kim & Park, 2017) $(Kim & Park, 2017)$ | | | | |
| 27 | Haabari villaga | (Kim & Park 2017) | | | | |
| 20 | lijok Gaet village | (Kim & Park, 2017) $(Kim & Park, 2017)$ | | | | |
| 30 | Sea Canoe company | (Kontogeorgopoulos, 2005) | | | | |
| 31 | Chambok CBET program | $(N_{Vaupapa} \& Thana 2004)$ | | | | |
| 32 | Giant panda habitats in Sichuan Province | $(M_{2} \text{ et al} 2019\text{ b})$ | | | | |
| 32 | Malaalaa lodga and pony trakking centra | (Maarns, 20170) | | | | |
| 37 | Malcalea louge and poly-tickking centre Mesomagor community | (Menseh, 2017) | | | | |
| 25 | WIL DTPUST and Cumbi community | (Musavangana, 2017) | | | | |
| 35 | Ololosokwan villago | (Nalson 2004) | | | | |
| 30 | Sinva villago | (Nelson, 2004) | | | | |
| 20 | Jaka Natron | (Nelson, 2004) | | | | |
| 20 20 | Lake Nation | (Neison, 2004) | | | | |
| 39 40 | Las Marias, recir community | (Nielsen, 2001) | | | | |
| 40 | Ploplaya, Carifuna community | (Nielsen, 2001) | | | | |
| 41 | Tolado Esotouriam Association (TEA) | (Timethy & White 1000) | | | | |
| 42 | Toledo Ecolourism Association (TEA) | (Timotny & White, 1999) (Deimer & Welter, 2012) | | | | |
| 43 11 | The Chiphat research site | (Kellifer & Walter, 2013) (Sakata & Deidagur, 2012) | | | | |
| 44 | Grass-roots CBE1 project on Fergusson Island | $(Sakala \propto Prideaux, 2013)$ | | | | |
| 45 | Chaba Englose Community Prest | (Iviona Snanwania et al., 2013) | | | | |
| 40 | Choose Enclave Community Trust | (Stone, 2015) | | | | |
| 4/ | Posada Amazonas | (A. Stronza, 2005) | | | | |
| 4ð 40 | Koli i do INOI Simphoni villogo | (Walter, 2009) | | | | |
| 49 | Sirubari village | (watter et al., 2018) | | | | |
| 50 | Gnalegaun village | (watter et al., 2018) | | | | |

Table 17. The selected cases for the case study meta-analysis, based on the Case Study Selection Criteria.

APPENDIX III

Table 18. Overview of used values to display the boxplots. These were found by the scores defined by the coding process.

| Boxplot | Frequency | Frequency Codes | | | | Frequency Values | | | |
|-------------|-----------|-----------------|-----|------|-----|------------------|------|-----|--|
| elements | CSFs | (1) | (0) | (-1) | (X) | (+) | (+-) | (-) | |
| Minimum | 3 | 3 | 0 | 0 | 0 | 0 | 0 | 3 | |
| Quartile Q1 | 17 | 8 | 2 | 1 | 1.5 | 1 | 6.5 | 4.5 | |
| Quartile Q2 | 23 | 16 | 4 | 2 | 4 | 2 | 10 | 7 | |
| Quartile Q3 | 35.5 | 22 | 5.5 | 3.5 | 6 | 4 | 16 | 8 | |
| Maximum | 43 | 31 | 8 | 9 | 8 | 19 | 26 | 14 | |
| Mean | 25.9 | 15.9 | 3.9 | 2.5 | 3.6 | 3.7 | 11.5 | 7.1 | |
| IQR* | 18.5 | 14 | 3.5 | 2.5 | 4.5 | 3 | 9.5 | 3.5 | |
| Outliers | No | No | No | Yes | No | Yes | No | No | |

* = INTERQUARTILE RANGE